

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Šcasný, Milan; Massetti, Emanuele; Melichar, Jan; Carrara, Samuel

Working Paper Quantifying the Ancillary Benefits of the Representative Concentration Pathways on Air Quality in Europe

Nota di Lavoro, No. 84.2015

Provided in Cooperation with: Fondazione Eni Enrico Mattei (FEEM)

Suggested Citation: Šcasný, Milan; Massetti, Emanuele; Melichar, Jan; Carrara, Samuel (2015) : Quantifying the Ancillary Benefits of the Representative Concentration Pathways on Air Quality in Europe, Nota di Lavoro, No. 84.2015, Fondazione Eni Enrico Mattei (FEEM), Milano

This Version is available at: https://hdl.handle.net/10419/130270

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



NOTA DI LAVORO 84.2015

Quantifying the Ancillary Benefits of the Representative Concentration Pathways on Air Quality in Europe

Milan Ščasný, Charles University in Prague, Environment Center Emanuele Massetti, Georgia Institute of Technology, CESIfo and FEEM Jan Melichar, Charles University in Prague, Environment Center Samuel Carrara, Fondazione Eni Enrico Mattei

Climate Change and Sustainable Development Series Editor: Carlo Carraro

Quantifying the Ancillary Benefits of the Representative Concentration Pathways on Air Quality in Europe

By Milan Ščasný, Charles University in Prague, Environment Center Emanuele Massetti, Georgia Institute of Technology, CESIfo and FEEM Jan Melichar, Charles University in Prague, Environment Center Samuel Carrara, Fondazione Eni Enrico Mattei

Summary

This paper presents estimates of the economic benefit of air quality improvements in Europe that occur as a side effect of GHG emission reductions. We consider three climate policy scenarios that reach radiative forcing levels in 2100 of three Representative Concentration Pathways (RCPs). These targets are achieved by introducing a global uniform tax on all GHG emissions in the Integrated Assessment Model WITCH, assuming both full as well as limited technological flexibility. The resulting consumption patterns of fossil fuels are used to estimate the physical impacts and the economic benefits of pollution reductions on human health and on key assets by implementing the most advanced version of the ExternE methodology with its Impact Pathway Analysis. We find that the mitigation scenario compatible with $+2^{\circ}$ C reduces total pollution costs in Europe by 76%. Discounted ancillary benefits are more than €2.5 trillion between 2015 and 2100. The monetary value of reduced pollution is equal to €22 per abated ton of CO2 in Europe. Less strict climate policy scenarios generate overall smaller, but still considerable, local benefits (14 € or 18 € per abated ton of CO2). Without discounting, the ancillary benefits are in a range of ≤ 36 to ≤ 50 per ton of CO2 abated. Cumulative ancillary benefits exceed the cumulative additional cost of electricity generation in Europe. Each European country alone would be better off if the mitigation policy was implemented, although the local benefits in absolute terms vary significantly across the countries. We can identify the relative losers and winners of ancillary benefits in Europe. In particular, we find that large European countries contribute to as much as they benefit from ancillary benefits. The scenarios with limited technology flexibility do deliver results that are similar to the full technology flexibility scenario.

Keywords: Ancillary benefits, External costs, Climate change mitigation, Integrated Assessment Models, ExternE, Impact Pathway Analysis

JEL Classification: Q47, Q51, Q53, Q54

Address for correspondence Milan Ščasný Jose Martiho 2 162 00 Prague 6 Czech Republic E-mail: milan.scasny@czp.cuni.cz

Quantifying the ancillary benefits of the Representative Concentration Pathways on air quality in Europe

Milan Ščasný^{**}, Emanuele Massetti[#], Jan Melichar^{*}, Samuel Carrara[~]

Abstract

This paper presents estimates of the economic benefit of air quality improvements in Europe that occur as a side effect of GHG emission reductions. We consider three climate policy scenarios that reach radiative forcing levels in 2100 of three Representative Concentration Pathways (RCPs). These targets are achieved by introducing a global uniform tax on all GHG emissions in the Integrated Assessment Model WITCH, assuming both full as well as limited technological flexibility. The resulting consumption patterns of fossil fuels are used to estimate the physical impacts and the economic benefits of pollution reductions on human health and on key assets by implementing the most advanced version of the ExternE methodology with its Impact Pathway Analysis. We find that the mitigation scenario compatible with +2°C reduces total pollution costs in Europe by 76%. Discounted ancillary benefits are more than €2.5 trillion between 2015 and 2100. The monetary value of reduced pollution is equal to \notin 22 per abated ton of CO₂ in Europe. Less strict climate policy scenarios generate overall smaller, but still considerable, local benefits (14 € or 18 € per abated ton of CO₂). Without discounting, the ancillary benefits are in a range of €36 to €50 per ton of CO₂ abated. Cumulative ancillary benefits exceed the cumulative additional cost of electricity generation in Europe. Each European country alone would be better off if the mitigation policy was implemented, although the local benefits in absolute terms vary significantly across the countries. We can identify the relative losers and winners of ancillary benefits in Europe. In particular, we find that large European countries contribute to as much as they benefit from ancillary benefits. The scenarios with limited technology flexibility do deliver results that are similar to the full technology flexibility scenario.

Keywords: ancillary benefits; external costs; climate change mitigation; Integrated Assessment Models; ExternE; Impact Pathway Analysis

JEL: Q47, Q51, Q53, Q54

^{*} Charles University in Prague, Environment Center, Prague, Czech Republic

[#]Georgia Institute of Technology, CESIfo and FEEM

[~] Fondazione Eni Enrico Mattei, Milan, Italy

[°] Corresponding author. Contact: Jose Martiho 2, 162 00 Prague 6, Czech Republic; <u>milan.scasny@czp.cuni.cz</u>.

1 Introduction

There is wide consensus on the need to regulate Greenhouse gases emissions (GHG) to limit the adverse impact of climate change. Hardly any country in the world disputes that the current pattern of GHG emissions is unsustainable. However, there is wide disagreement on the magnitude of the optimal emission reductions, on the distribution of effort across countries, and on the timing of the emission reductions. This is hardly a surprise: the climate problem is a perfect example of intragenerational and inter-generational externality. Emission abatement costs are local, while the benefits of controlling the increase of GHG concentrations are global and will be experienced only far in the future.

Current carbon taxes and permit trading programs cover only around 12 percent of global emissions (Parry 2014) and current policies are still far from being effective and optimal (see, for instance, Máca et al. 2012; Somanathan et al. 2014). It is unclear if future negotiations will be able to change this dismal state of climate policy. The chances are that if any agreement can be reached, it will entail a low level of commitment (Barret and Stavins 2003).

Some authors suggest that one way to kick-start climate policy is to think locally instead of globally.¹ They notice that climate mitigation policy may have immediate and local co-benefits that could partially or totally offset the cost of reducing GHG emissions. The question then becomes: "what scale of CO₂ pricing is in countries' own interest" (Parry et al. 2014).

For example, investment in low-emission technologies may spur innovation and growth in markets with unemployment and knowledge externalities (Bauer et al. 2009; Acemoglu et al. 2012; OECD 2012a). Carbon taxes or revenues from auctioning emission allowances may reduce distortions in the tax system if appropriately managed (Pearce 1991; Goulder 1995; Carraro et al. 1996; Bovenberg 1999).

As a large fraction of GHG emissions comes from the combustion of fossil fuels, the use of oil, natural gas and coal must be substantially reduced to meet any long-term climate mitigation goal. Fossil fuels also cause local and present environmental damage because, if burnt, they release pollutants that negatively affect human health, ecosystems and other assets. Climate mitigation policies would thus reduce the burden of ground-level air pollution. The reduction of fossil fuels consumption and the shift of the energy mix away from coal – the most polluting among the fossil fuels – to natural gas is expected to reduce the concentrations of particulate matters, sulphur dioxide, nitrogen oxides, ozone precursors (NO_X and VOC) and other toxic pollutants. Climate change mitigation can thus generate immediate ancillary benefits² by reducing the negative impacts of pollution on human health, crop yields, building materials or ecosystems.

The aim of this paper is to provide estimates of local air quality ancillary benefits of carbon taxes in Europe. We assume that the carbon price is exogenously set to achieve the most recent set of

¹ For a survey see Chapter 15 of the Fifth Assessment Report of Working Group III to the IPCC (Somanathan et al. 2014).

² The Intergovernmental Panel on Climate Change (Third Assessment Report) distinguishes between ancillary benefits and co-benefits (IPCC 2001). Ancillary benefits are related to policies or measures that are targeted entirely on climate change mitigation, while co-benefits are referred to when policies or measures are designed for more than one target (Dudek et al. 2003). We consider aggregated ancillary benefits in this report.

climate mitigation targets used in the literature³ and we estimate the air-quality benefits of such climate policies. We argue that these benefits should be kept logically distinct from the benefits of climate mitigation. Our goal is to estimate the size of local air quality externalities compared to the global carbon price. We also estimate the value of ancillary benefits per ton of GHG emission abated and per Euro of abatement costs in the power sector in Europe.

In order to assess the local environmental benefits of climate policy we use the integrated assessment model WITCH to provide fossil fuel use in the power sector under alternative carbon tax scenarios and an impact pathway analysis embedded in the ExternE method (EC 2005) to estimate the economic damage of emissions of particulate matters or various fractions (PM10, PM2.5), sulphur dioxide (SO₂), nitrogen oxides (NO_x), non-methane volatile organic compounds (NMVOCs), and heavy metals (As, Cd, Cr, Hg, Ni, Pb). Although our economic model is global, we consider the ancillary benefits for Europe and the abatement costs involved for the electricity sector in Europe. Despite recent efforts to increase the level of renewable electricity generation, coal fired power plants still provide about 25% of total electricity generation in Europe. Coal is the fossil fuel that causes the largest local and global environmental damage. Any effort to reduce GHGs emissions will also yield large local environmental benefits in Europe.

Overall the strengths of our approach are: (1) cost effective energy and emissions scenarios that deliver the most recent climate mitigation targets; (2) the use of atmospheric circulation models to track the impact of air pollution using; (3) the ability to estimate EU-wide and country-level ancillary benefits; (4) the inclusion of Eastern European countries, typically not considered by European studies; (5) the possibility to separate the ancillary benefits between those caused by reduction in local energy use and those due to emissions reductions elsewhere.

In the RCP2.6 mitigation scenario – a +2°C compatible scenario – we find that discounted total ancillary benefits for Europe are greater than \pounds 2.5T, or about \pounds 21.6 per abated ton of CO₂. Less strict climate policy scenarios generate overall smaller local benefits, but the magnitude of ancillary benefits per abated ton of CO₂ is still considerable (\pounds 14.4, or \pounds 18.3, respectively) than under stricter mitigation. Without discounting, the ancillary benefits are in a range of \pounds 36 to \pounds 50 per ton of CO₂ abated. Over the whole century ancillary benefits per abated ton of CO₂ are 7-times larger than the carbon price in the less aggressive climate scenario and only 0.06 of the carbon price that has quite large value in the RCP2.6 mitigation scenario. Ancillary benefits cumulated over the century exceed cumulative additional cost of electricity generation in Europe in both strict mitigation scenarios. About 73% of total ancillary benefits attributable to the RCP4.5 mitigation scenario have domestic origin, while the rest of benefits is enjoyed in other countries than in a country where the emissions are abated. There are relative losers and winners with respect to produced and received ancillary benefits. However, there is no country in Europe that will be worse-off if the mitigation policy was implemented. Scenarios with technological constraints do not lead to qualitatively different results.

The rest of this paper is structured as follows: Section 2 reviews the literature on the ancillary benefits, while Section 3 describes the integrated assessment model, the impact pathway analysis, and the linkages between the two approaches. Section 4 describes climate policy scenarios and energy use in Europe. Section 5 presents estimates of the side benefits of climate policy. Conclusions follow.

³ We conduct cost-effectiveness analysis in our study rather than cost-benefit analysis in order to provide information on the efficient level of regulation and hence the optimal level of carbon tax.

2 Literature Review

The ancillary benefits of GHG mitigation policies have been investigated in a variety of empirical studies. The literature has used a wide range of techniques and models, it has covered different geographic areas, pollutants and pollution-related impacts. Some studies rely on air quality modelling while others use default damage factors per pollutant.⁴ Due to these differences, results from these studies are difficult to compare.

Among the studies that analyze the effect of a policy on multiple pollutants, a first distinction can be made between studies that focus on physical impacts (e.g. Meyer et al. 1998) and studies that emphasize the monetization of the impacts (e.g. Burtraw et al. 2003). Several studies investigated the links between regional air pollution and climate policy in Europe (Syri et al. 2001; Alcamo et al. 2002; van Harmelen et al. 2002). Most, if not all, studies on ancillary benefits either utilize a damage factor for each pollutant or connect emissions to changes in concentrations, human exposures, physical effects and monetary damages using an integrated assessment model, such as the US APEEP (Muller and Mendelsohn 2007) or the EU ExternE's impact pathway analysis (Holland et al. 2011).

Regarding the economic modelling, the ancillary benefits are quantified (a) relying on a linear programming partial equilibrium energy model (e.g., Burtraw et al. 2003; Van Vuuren et al. 2006; Riekkola et al. 2011; Holland et al. 2011; Rečka and Ščasný 2013) that can be further enriched, for instance, by econometric estimates of abatement costs (Burtraw et al. 2014), or (b) using a computable general equilibrium model (e.g., Glomsrød et al. 1992; Scheraga and Leary 1993; Paltsev et al. 2005; Grossman et al. 2011; Nam et al. 2013; Kiuila et al. forthcoming). Bollen et al. (2009) use the MERGE model, a top-down optimization model with a good energy sector detail. MERGE is the model used in the ancillary benefits literature that is most similar to WITCH. There are however several differences between our study and Bollen at al. (2009). First, we are concerned with the estimation of ancillary benefits rather than the joint optimal determination of local pollution abatement and global GHG emissions reduction. Second we consider the most recent scenarios in the climate change literature and we provide information on ancillary benefits associated to each scenario. Third, while Bollen et al. (2009) only consider PM pollution, we include a much larger set of pollutants in our analysis. Fourth, we do not have a hard-linked air pollution impact module, which is a weakness, but the flexibility of a soft-link approach allows us to use a more complicated modelling approach that takes into account the atmospheric circulation of local pollutants. Finally, Bollen et al. (2009) deal with global costs and benefits while we provide EU-level and country-level analysis.

The range of estimates of air-quality ancillary benefits is large and depends on many factors, including climate policy scenario, modelling assumptions and the time period over which the benefits are calculated. In their review of the literature Davis et al. (2000) report a range from €0.6 to €78 (Dessus and O'Connor 1999) and €148 (Aunan et al. 2000) per ton of reduced CO₂ emissions.⁵ A review of 48 peer reviewed studies by Nemet et al. (2010) provides a range of the air quality cobenefits of climate change mitigation from €1.6 to €152 per ton of abated CO₂ with a mean of $€38/tCO_2$. Focusing on the co-benefits from mitigation in the US electricity sector, Nemet et al. provide a range of estimates ranging from €3 to €90, and found larger ancillary benefit estimates for

⁴ See, for instance, Bell et al. (2008) that discusses the methodological aspects in quantification of ancillary benefits.

⁵ Davis et al. (2000) and then OECD (2002) report the ancillary benefits per ton of carbon in 1996 US\$. We use OECD CPI and purchasing power parity and express the benefits in 2005 Euro per ton of CO_2 . Following same approach, we recalculated the 2008 USD from Nemet et al. (2010) and the 2010 USD from Parry et al. (2014) in 2005 Euro.

developing countries (with a mean of \pounds 62) than for the developed countries (mean \pounds 33). Parry et al. (2014) derive 'nationally efficient carbon prices' that reflect domestic non-internalized environmental benefits for top 20 emitters of GHG emissions. The efficient prices reflect primarily health co-benefits from reduced air pollution at coal plants and reductions in automobile externalities, which are net of fuel taxes or subsidies. They find that the cross-country average of nationally efficient prices is equal to \pounds 44 per ton of CO₂ in 2010 (with a range between \pounds 22 to \pounds 66 per ton as the damage values are reduced and increased by 50 percent). These prices substantially vary, however, across countries; the nationally efficient CO₂ price is as high as \pounds 224 per ton in Saudi Arabia, \pounds 48 per ton in China, or \pounds 28 in the United States. Brazil on average overcorrects for cobenefits through pre-existing policies that results in the negative efficient price of - \pounds 18. Coal air pollution damage is estimated for the top twenty emitters at \pounds 66 per ton. It is important to note that Parry et al. (2014) do not consider any benefit from reduced global warming when they calculate the 'nationally efficient carbon price'. They calculate the value of the ancillary benefit of a carbon tax and find the level at which the tax is equal to the local benefits.

For the EU, Holland et al. (2011) estimate that the 2°C stabilization scenario would reduce SO₂ emissions by 60%, NOx emissions by 46% and particulate matters by 19% using the partial equilibrium energy model GAINS. These emission reductions would lead to large health improvements and important co-benefits in ecosystems. The air quality co-benefits are estimated at \notin 43B per year by 2050 in the EU27, which corresponds to about \notin 24 per ton of CO₂. Markandya et al. (2009) obtained similar results for the EU, but much greater co-benefits in fast growing countries such as China and India. Barker and Rosendhal (2000) estimate the co-benefits from SO₂, NOx and PM10 reduction as an effect of carbon tax for Western Europe at \notin 41 per ton of CO₂ abated.

While the literature on ancillary benefits has considerably grown during the past twenty years, studies aiming at developing countries or transforming economies in Eastern Europe are relatively few (Morgenstern 2000). Aaheim et al. (1997) and Aunan et al. (2000) investigated the ancillary benefits of several policies, covering energy efficiency or public transport, in Hungary. They estimated annual health benefits in a range of \$370 to \$1,168m. Dudek et al. (2003) provide an analysis of ancillary benefits of energy market reforms and emission trading for Russia and observe 30,000 to 40,000 lives saved that are then monetized using the Value of Statistical Life (VSL). Using a linear optimization energy model linked to ExternE's impact pathway analysis, Rečka and Ščasný (2013) estimated the ancillary benefits of the EU Emission Trading Scheme (EU-ETS) in the Czech power sector at €3,100B during 2006-2030, or €4,100B if no new nuclear plant is allowed. These results imply the ancillary benefits of ≤ 15 per ton of CO₂ in both scenarios. Using same approach, Ščasný and Rečka (forthcoming) obtained the ancillary benefits for tightening CO₂ target in Slovakia at €11 per ton of CO₂. Kiuilia et al. (forthcoming) estimated co-benefits of full internalization of external costs attributable to local pollutants in the Czech Republic by CGE model in a range of €32 to €72, depending on the scenario and taxed sectors. The ancillary benefit estimates are summarized in Table A - 2 in the Appendix.

3 Modelling framework

3.1 The WITCH model

WITCH (World Induced Technical Change Hybrid) is an Integrated Assessment Model (IAM) designed to study the socioeconomic impacts of climate change and the implications of mitigation policies on the energy sector, the economy and climate. WITCH is a global model where countries of the world are grouped into thirteen regions: USA (United States of America), WEURO (Western EU and EFTA countries), EEURO (Eastern EU countries), KOSAU (South Korea, South Africa and Australia), CAJAZ

(Canada, Japan and New Zealand), TE (Transition Economies, namely Russia and Former Soviet Union states and non-EU Eastern European countries), MENA (Middle East and North Africa), SSA (Sub-Saharan Africa except South Africa), SASIA (South Asian countries except India), EASIA (South-East Asian countries), CHINA (People's Democratic Republic of China and Taiwan), LACA (Latin America and Central America) and INDIA (India).⁶ These regions strategically interact following the rules of an open-loop Nash game: each region maximizes its own welfare given the behaviour of all other regions. A cooperative solution, where one global social planner jointly maximizes a social welfare function can also be implemented but was not used to generate the scenarios used in this study.

The model is defined as hybrid because it features an aggregated top-down Ramsey-type optimal growth model combined with a detailed description of the energy sector. The aggregated economic model is structured according to a Constant Elasticity of Substitution (CES) framework where the two macro-inputs, capital and labour, are combined with energy to produce the final output. The energy node is then disaggregated in a detailed, fully integrated, bottom-up section that features all major power technologies and non-electric energy demand aggregated by fuel. The model is able to study the evolution of the energy sector in relationship with major economic and climate variables.

Technical change is endogenous in WITCH and is modelled via Learning-by-Researching (LbR) and Learning-by-Doing (LbD) effects. LbR determines technology cost reduction by means of dedicated investments in R&D capital. International R&D spillovers are also taken into account. LbD reduces the investment cost of renewable and backstop energy technologies as a consequence of progressive deployment of the technology. A more detailed description of the model can be found in Bosetti et al. (2006 2007 and 2009).

3.2 The ExternE method

We estimate the impact of climate change mitigation on air quality using a method based on the most recent ExternE (Externalities of Energy)⁷ Impact Pathway Analysis (IPA).⁸ The IPA is an analytical procedure examining the sequence of processes through which polluting emissions result into external damages. The method allows to estimate the marginal physical impact and the marginal cost of pollution from each power plant (in general from any emission source), as a function of the technology and of the location of the plant.

The IPA comprises four basic steps: (i) selection of the reference power plant, determination of the technology used and of the harmful emissions released, (ii) calculation of changes in pollutant concentration for all affected regions using atmospheric dispersion models, (iii) estimation of

⁶ For the purposes of this work, the European Union is given by the sum of WEURO (Western Europe) and EEURO (Eastern Europe), although this is not rigorously correct due to the presence of the EFTA countries in the EU.

⁷ The European Commission in collaboration with the US Department of Energy launched a joint research projects to assess the energy-related externalities in 1991 (European Commission 1995; ORNL and RFF 1995). Following a detailed bottom-up methodology relying on impact pathway approach, the EU/US studies provided estimates of marginal external costs of electricity production from a wide range of energy technologies at various locations. The EC provided additional funding over the years to improve the ExternE accounting framework and to expand it to new EU member states and to other non-EU countries. The ExternE IPA framework that we use has been recently updated within the NEEDS project (<u>http://www.needs-project.org/</u>). For more information on ExternE see <u>http://www.externe.info</u>. Weinzettell et al. (2012) apply the ExternE method to quantify production and consumption related externalities of power sector in Europe.

⁸ An internet accessible version of EcoSense (EcoSenseWeb1.3) was developed within the NEEDS project (Preiss and Klotz 2008).

physical impacts from exposure using concentration-response functions (CFRs), and (iv) economic valuation of impacts using direct costs (effect on crop yield, damage on building materials or biodiversity) or compensating/equivalent surplus measured through the willingness-to-pay approach. The ExternE's IPA method is very similar to an integrated assessment model used in the American studies to connect emissions to changes in concentrations, human exposures, physical effects and monetary damages by the Air Pollution Emission Experiments and Policy model (APEEP, see Muller and Mendelsohn 2007, 2009; Muller et al. 2011; Grossman et al. 2011).

The IPA procedure has been incorporated into EcoSense, the integrated atmospheric dispersion and exposure assessment model that we use for our analysis.⁹ EcoSense uses air transport models to control changes in the atmospheric concentration of pollutants at local, regional and global level.¹⁰ The model then determines a range of impacts on human health, buildings, biodiversity, and crop yields using concentration-response functions. We evaluate the economic impact of micro-pollutants using generic estimates of marginal costs – i.e. the same damage value regardless which country releases the micro-pollutant – as estimated in the ExternE project series. The loss of ecosystems is assessed using a measure of Potential Disappeared Fraction of species (Frischnecht and Steiner 2006) linked to acidification and eutrophication. We use appropriate concentration-response functions to estimate the economic loss from mortality and morbidity, from agricultural productivity losses and for damages to building materials. Valuation methods of welfare economics are used to translate the physical impacts into monetary impacts.

Impacts on human health, mainly on mortality, are the most important among all impacts. In order to establish a causal relationship between pollution and human morbidity and mortality, ExternE uses concentration-response functions calibrated using a large number of epidemiological and toxicological studies. At the beginning of the ExternE project the CRFs of all European countries were calibrated using studies for the United States. The European functions have now been re-calibrated using epidemiological and toxicological studies for Europe. The economic loss due to increased mortality is estimated using the Value of Life Year (VOLY) (Desaigues' et al. 2011), reflecting recent changes in the ExternE methodology. Previously, ExternE used a uniform Value of Statistical Life (VSL) to value excess mortality. Several studies have argued that the VSL is appropriate to value large losses of life expectancy from fatal accidents but it should not be used to estimate the usually smaller impact of pollution on life expectancy, especially of elderlies (Rabl et al. 2014). Regardless which one of the two metrics is used, they should be both based on the willingness to pay for a small reduction in risk of dying (Hammitt 2007).

In this study, following the ExternE method, the VOLY for chronic mortality is set at €40,000, the recommended value of ExternE for cost–benefit analyses of EU-level policies.¹¹

⁹ <u>http://ecosenseweb.ier.uni-stuttgart.de/</u>

¹⁰ EcoSense uses three models of air quality: (i) the Industrial Source Complex Model for transport of primary air pollutants on a local scale delaminated by 100 x 100 km around the power plant, (ii) the EMEP/MSC-West Eulerian dispersion model for modelling transport and chemical transformation of primary pollutants on a regional scale covering all Europe, and (iii) the N-Hemispheric Model which served for estimation of the intercontinental influence primary and secondary pollutants (secondary inorganic aerosols, tropospheric ozone).

¹¹ The recommended value of so called chronic VOLY is based on the mean estimate of the willingness to pay for changing life expectancy by two months using data from a pooled sample of nine European countries. Data is adjusted using a simple benefit transfer technique to correct for the differences in income and population in EU Member States. Monetary values for work loss day, medical costs, or the willingness to pay to avoid illnesses also reflect EU-wide averages.

The valuation of morbidity is by no means trivial. Morbidity increases medical costs and causes a loss of productivity, but it also causes large, harder to measure, disutility from pain and suffering. There has been recent interest in assessing the value of utility losses from illness, but more research is needed, especially to value the damages from chronic illness (e.g. chronic bronchitis or asthma symptoms). We follow here the valuation of additional morbidity proposed in the NEEDS update of ExternE. Their values range from €1 for each use of bronchodilator to €200,000 per new case of chronic bronchitis.

Crop losses are valued at the international market prices. The impacts on building materials are assessed using replacement and maintenance costs, the assessment of biodiversity impacts is based on restoration costs.

We use country-specific impacts valued by EU-wide social damage costs to express them in ≤ 2005 per ton of emission of pollutant as estimated by the project NEEDS. We include the most common air pollutants (SO₂, NOx, PM_{2.5}, PM₁₀, NMVOC) and heavy metals (As, Cd, Cr, Hg, Ni, Pb). NEEDS provides the average value of external costs per ton of emission for a total of 39 European and non-European countries.¹²

The marginal social cost of emissions of heavy metals is developed in NEEDS using data from the EUfunded project ESPREME (2007) and from two studies (Fantke 2008; Spadaro and Rabl 2007). Since the background concentration of NO_x , SO_2 , NMVOC and NH_3 influences the generation of secondary pollutants – e.g. ozone, sulphates, nitrates – the NEEDS project provides two sets of impact estimates for non-metals: one in 2010 and the other in 2020 (see, Preiss and Klotz 2008). The EU-wide average external cost of pollution is reported in Table 1.

Pollutant	PM _{2.5}	PM _{coarse}	NO _X	SO ₂	NMVOC	Cd	As	Ni	Pb	Hg	Cr
Emission scenario 2010	12.08	0.52	5.84	6.54	1.01	84.69	536	1.67	284	8371	9.90
Emission scenario 2020	11.10	0.43	6.00	6.55	0.52	74.59	472	1.47	251	7372	8.72

Note: Monetary values in \pounds 2005 per ton of emitted pollutant. PM_{coarse} indicates particulate matters with an aerodynamic diameter between 2.5 and 10 μ m. Source: Preiss and Klotz (2008).

Table 1. Average damage factors for air quality pollutants and heavy metals for the 2010 and 2020 background emission scenarios (thousands €2005 per ton).

As an example, Figure 1 presents the EU-wide impacts of one additional unit of $PM_{2.5}$ released in any EU country, in 2010 and in 2020. The figure also displays the EU average marginal cost of $PM_{2.5}$. The marginal cost of pollution in Figure 1 is the EU-wide social cost of pollution. The social cost differs across countries because of different environmental conditions, different density of receptors, and different characteristics of the receptors. For instance, the unit damage cost of $PM_{2.5}$ is much smaller (about 4,000 \in per ton) in countries with lower population density in their neighbouring countries (Finland, Norway, or Sweden) than in countries (Benelux, Germany) at the centre of a heavily populated region (about 20,000 \in per ton).

The cost of pollution for Europe is equal to the sum of the cost attributable to fuel use and hence emissions released from their burning in each of the 28 European countries. The released emissions

¹² The costs were estimated using several runs of the EcoSenseWeb tool with the EMEP/MSC-West Eulerian pollution dispersion model.

have an effect on receptors (humans, species, assets) in the 28 European countries but they also affect emission receptors in the rest of the world. Using the EcoSenseWeb tool, we derive the cost of one additional unit of pollution released in country *i* that is inflicted to each of the remaining European countries *j* and to the rest of the world (*j*+1). In our case, this procedure generates 28 x 29 pollution costs. We assume that the distribution of impacts due to emissions releases from country *i*, i.e. regional increase in ambient concentrations, remains same over the whole period 2015-2100 as in the reference year. In other words, we do not model meteorological conditions and transport of pollutants over time.



Source: Preiss and Klotz (2008).

Figure 1. Damage factors per ton of PM_{2.5} for 2010 and 2020 background emission scenario (thousands €2005 per ton).

3.3 Linking the WITCH model and the ExternE approach

The ancillary benefits of reducing GHG emissions are derived using data on fossil fuels consumption for power generation (*EN*) from WITCH. WITCH provides energy scenarios for Eastern and Western Europe until 2100. In order to provide country-level scenarios of fossil fuels combustion for power generation ($EN_{f,j}$) we assume that the current distribution of power generation across countries remains unchanged.¹³ Emission factors (*EF*) from fossil fuels used for power generation are from the EMEP EEA air pollutant emission inventory guidebook (EMEP/EEA, 2013) and are listed in Table A - 1 in the Appendix. The damage factors (*DF*) for each pollutant are from the Impact Pathway Analysis of

¹³ Primary energy use in country *j* equals to $(EN_f sh_j)$, where EN_f denotes primary energy use for electricity generation in one of the two European regions from WITCH and sh_j indicates the share of country *j* on use of fuel *f*, as it was in the base year 2005.

ExternE. We assume that damage factors change after 2015, following the assumptions in the EcoSenseweb software.

There are five channels through which both GHG and air quality emissions can be reduced: (1) reducing total economic output (scale effect), (2) restructuring the economy towards less emissionintensive sectors (composition effect), (3) reducing the fuel intensity of production (fuel intensity effect), (4) using fuels with reduced or no carbon emissions (fuel mix effect), and (5) utilizing more efficient end-of-pipe technologies (emission-fuel intensity effect).¹⁴ By linking WITCH and IPA we are able to take into account channel (1); channel (2) is also accounted for, but it cannot be separated from (3) because WITCH has only one aggregate final good sector; the mix of the capital-labour aggregate with energy is instead optimally determined in WITCH; channels (4) and (5) are also accounted for.

The change of the fuel mix responds to changes in the cost of fuels and of power generation capital and to the carbon tax penalty. However, WITCH does not track non-GHG emissions.¹⁵ As the non-GHG emission factors for fuels (EF_f) remain constant over the entire period, it implies time invariant efficacy of the end-of-pipe abatement technologies and hence constant emissions per unit of fuel over time.

As noted above, in ExternE the willingness to pay used to derive the damage factors reflects EU average values, implicitly assuming that all Europeans share the same preference for avoiding adverse health and other negative impacts.

The real value of pollution costs (PC) attributable to air polluting emissions released by country i in year t is derived as follows:

$$PC_{t,i} = \sum_{j} \sum_{f} \sum_{p} \left(EN_{j,f} \cdot EF_{f,p} \cdot DF_{j,p} \right) \cdot \left(\frac{Y_j}{Y_{EU}} \right)^{\varepsilon_y^{wtp}} \cdot \prod_{s=2005}^t \left(1 + g_s \cdot \varepsilon_y^{wtp} \right)$$
(1)

where *f* denotes the fuel type (coal, gas, oil, biomass, nuclear, renewable energy), *p* denotes the pollutant (SO₂, NOx, PM, NMVOC, heavy metals). For each country *i*, the cost of air pollution is determined summing damages in each own country and on all other '*victim*' countries *j*. We sum the damage of emissions on each other and 28 European countries plus the rest of Europe¹⁶, including *i*th country, in that airborne concentrations are changed due to emissions released from the emitting country *i*. Thus, the first term in equation (1) $(EN_{f,j} \cdot EF_{f,p} \cdot DF_{p,j})$ quantifies the external cost of non-GHG emissions released by country *i*. Our impact assessment follows a static approach: neither of the parameters on emission-fuel factors (EF_{f,p}), damage factors (DF_{p,j}), country's share on fuel use (sh_j), or $\varepsilon_y^{\text{wtp}}$ are time invariant. Then, the second and the third terms adjust for the differences in income levels across 28 European countries, and over time, respectively. These adjustments are based on a simple benefit transfer that assumes that richer people are willing to pay more – not necessarily strictly proportionally – than poor people. The second term translates the EU average value into the values that are more relevant to a '*victim*' country *j*, where *Y_j* and *Y_{EU}* are real per

¹⁴ Carbon capture and sequestration is an end-of-pipe technology for GHG emission reductions.

¹⁵ A new version of the model with local pollutants and other non-GHG emissions was under preparation while this article was written.

¹⁶ To analyse the distribution of the impacts, country-specific damage factors and external costs are derived for 28 countries. We have data on the EU28 countries, with the exceptions of Malta and Croatia. We include also Norway and Switzerland. The impacts on the rest of the world are valued as if they were born in the emitting country.

capita GDP in purchasing power standards in country *j*, or in the EU, respectively, in the year 2005. As a result we monetize physical impacts by country-specific values, instead of using an EU-average value (e.g. Holland et al. 2011). The third term adjusts for possible changes in income over time in the 'victim' country. ε_y^{wtp} denotes the income elasticity of willingness to pay, for which we use a value equal to 0.8, similarly (OECD 2012b, 2014; WHO and OECD 2015).¹⁷ The parameter g_p is real growth in per capita GDP, as endogenously determined by WITCH for the two European regions. No equity weighting is assumed in our calculation of pollution costs.

Several adjustments are made to ensure the comparability of monetary values and to guarantee consistency between the two modelling approaches. The WITCH model provides results in 2005 USD, while the pollution costs in the ExternE are expressed in 2000 €. All values are converted to 2005 € by using the GDP deflator and the market exchange rate for 2005. In order to compare the ancillary benefits with economic impacts expressed in present value, the ancillary benefits are discounted using the interest rate endogenously determined in WITCH.¹⁸

We repeat this exercise for a set of climate mitigation scenarios recently developed. We follow Riekkola et al. (2011) and express the ancillary benefits as the difference between the air pollution damages in the baseline scenario and in the policy scenario. For each policy scenario we calculate the absolute benefit of pollution reduction, the benefit per ton of CO_2 emissions abated in the European electricity sector, the ratio between air pollution ancillary benefits and total CO_2 abatement cost in the electricity sector, and the benefit as a ratio of the global carbon price.

4 Scenarios

In this study we use four climate mitigation policy scenarios developed using WITCH for the EUfunded project GLOBAL-IQ (Massetti et al. 2014). The Reference scenario assumes the continuation of observed trends. We assume that there is no policy to reduce GHG emissions in the Reference scenario. We then use three climate policy scenarios in which emissions decline over time to achieve three levels of radiative forcing in 2100: 6.0, 4.5 and 2.6 watts/m² (they correspond to concentrations equal to 850, 650, and 490 ppm CO_2eq). These scenarios are named RCP6.0, RCP4.5 and RCP2.6 because the radiative forcing levels are those used in the Representative Concentration Pathways (RCP) (Van Vuuren et al. 2011).

The RCPs have been recently developed to provide greenhouse gases emissions scenarios to the climate models that have been used to generate the most recent set of climate change scenarios (Stocker et al. 2013). The RCPs are the new standard in the climate change literature and are now being complemented by a set of socio-economic scenarios, the Shared Socioeconomic Pathways (SSP) (Ebi et al. 2014, Van Vuuren et al. 2014).

¹⁷ The OECD (2012b) review finds that the income elasticity of the VSL is in the range of 0.7 and 0.9 in most of the regressions that use screening criteria. In other studies this range is substantially lower – about 0.3 to 0.4. Viscusi (2000) finds studies that use a value greater than unity. In most studies the income elasticity of the VSL ranges between 0 and 1 and the income elasticity of WTP around unity may be justified for the transfers between countries with heterogeneous income (Czajkowski and Ščasný 2010).

¹⁸ In WITCH the pure rate of time preference declines over time. It starts at 3 % p.a. in 2005 and declines to about 2 % p.a. in 2100. The interest rate of the economy declines over time following the Euler equation. The model is calibrated so that developed regions have an interest rate equal to about 5% per year in 2005 while developing regions have an interest rate equal to 7% per year in 2005.

Each RCP scenario may be the outcome of different socioeconomic pathways. For example, a high emission scenario may be the outcome of both a fast-growing but highly efficient global economy and of a sluggish and inefficient global economy. Analogously, the same emission trajectory (RCP) may be consistent with both high and low global economic inequalities.

The Reference scenario used in this study reproduces population and economic growth of the SSP2 scenario. The SSP2 is a central-case scenario because current trends are assumed to continue indefinitely in the future. The SSP2 is commonly identified as the "Middle of the road" scenario.¹⁹

In our Reference scenario population and economic growth follow the trends observed in the past. There are considerable energy efficiency improvements in the Reference scenario, but global energy demand substantially increases as a result of economic and population growth. Without any global agreement to reduce greenhouse gases emissions this incremental energy demand is mainly covered using fossil fuels. The carbon dioxide emissions from the use of fossil fuels are the main source of additional GHGs that are released in the atmosphere in the Reference scenario. The power sector is one of the largest contributors to CO_2 emissions. In 2010 about 10 Gt CO_2 are released from power plants at global level, of which 1.4 Gt originate from Europe. This is equivalent to about 30% of total CO_2 emissions. The carbon intensity of the European power sector declines up to 2030 and then it climbs again as coal re-gains a share of the power mix. In 2050 emissions are equal to 1.1 Gt CO_2 per year (24% of European CO_2 emissions) and in 2100 emissions from the power sector total 1.7 Gt CO_2 per year (35% of European CO_2 emissions) (see Figure 2).

The climate module used by WITCH indicates that radiative forcing achieves 6.6 W/m^2 in 2100 in the Reference scenario. The global mean temperature increases by 4.1 °C in 2100, with respect to the pre-industrial level.

Emissions sharply decline in the RCP scenarios. While the RCP6.0 is only marginally different from the SSP2 scenario, the RCP4.5, and especially the RPC2.6 scenarios indicate that it is optimal to drastically cut emissions from power generation. Coal and natural gas power plants are retrofitted with carbon capture and storage equipment while investments in wind and solar power increase. Investments in end-use energy efficiency R&D reduce the demand of energy and of electricity in particular. As early as in 2040 emissions in the RCP2.6 scenario become negative because biomass (carbon neutral) is burnt in power plants with carbon capture and storage. Thus the power sector quickly shifts from being a major source of emissions to being a net sink.

¹⁹ The Reference scenario is thus characterized by: (1) slowly decreasing fossil fuel dependency, (2) reductions of resource and energy intensity, (3) uneven development of low-income countries, (4) weak global institutions, (5) slow continuation of globalization, with some barriers remaining, (6) well regulated information flow, (7) medium economic growth, slow convergence, (8) high intra-regional disparities, (9) medium population growth related to medium educational investments, (10) delay of achievement of the Millennium Development Goals (MDGs).



Figure 2. Global and European carbon dioxide emissions from fossil fuels used in the power sector.

All the RCP scenarios assume full technological flexibility. For example carbon capture and sequestration is deployed on a large scale starting from 2020. Biomass is assumed to be widely available. Energy efficiency gains, although costly, greatly contribute to the reduction of electricity demand. The energy system as a whole is assumed to quickly adapt to the new regulatory regime. In order to test the effect of a less flexible technological setup on local pollution we also use three scenarios in which technological adaptation is limited (see Massetti et al. 2014, Leimbach et al. 2014). These are limited adaptation scenarios (LA). We calculate the new carbon price that is consistent with the RCP4.5 scenario and we assess how the technology mix and the distribution of emissions changes across technologies and across sectors (total emissions are unchanged because we still impose the long-term climate target to be achieved). Specifically, we consider three additional scenarios: with limited energy efficiency (LA-EE), limited renewable energy (LA-REN) and limited supply and trade of biomass (LA-BIO). For the limited adaptation scenarios we calculate emissions reductions and climate mitigation policy costs using a 'limited adaptation' Reference scenario with the same technology constraint as in the policy scenario. We are particularly interested in whether there is any meaningful difference in local pollution as a consequence of constraints to key mitigation technologies.

5 Results

5.1 Economic impacts

In all climate mitigation scenarios our policy tool is a uniform – over countries, sectors and GHG emissions – carbon tax. The time path of the carbon taxes is shown in Figure 3.²⁰ The policy starts in 2015 and with the RCP6.0, RCP4.5 and RCP2.6 scenarios the global mean temperature increases by 3.7° C, 3.0° C and 2.0° C, respectively. In the RCP6.0 case, the carbon tax starts at 3 USD/tCO₂eq and achieves 14 USD/tCO₂eq in 2100. The level of the tax is very low because the RCP6.0 target is not far from the Reference scenario. The carbon tax for the RCP4.5 reaches $335/tCO_2$ eq USD in 2100. Due to limited technology adaptation, the level of the carbon tax is 10% higher in LA-BIO and by 31% higher in LA-EE in 2100, but it is almost identical in LA-REN. The variations of the carbon tax reflect the relative importance of alternative mitigation channels. The tax escalates in RCP2.6 and reaches more than 3,500 USD/tCO₂eq in 2100. The tax is recycled lump-sum in each region. Nothing but the price of carbon is changed with respect to the corresponding reference scenarios.

²⁰ The three carbon tax trajectories are consistent with the radiative forcing targets. They are not socially optimal taxes because they are obtained solving the model in the cost-effectiveness mode.



Notes: RCP6.0 on the secondary vertical axis.

Figure 3. The carbon tax – policy scenario.

As it becomes more and more expensive to emit GHG emissions, the model scenario projects investment in carbon-free technologies for the energy sector, in energy efficiency R&D, in R&D to develop carbon-free backstop technologies, it substitutes the energy input with capital and labour and it invests in a series of activities to reduce CO₂ emissions from deforestation and other non-CO₂ gases. The resulting emissions trajectories are the optimal (efficient) solution of a complex inter-temporal and strategic optimization problem. Feedback from technology spillovers and from global energy markets are internalized in the solution of each regional social planner.

Emissions reductions in the power sector are the result of reduced demand due to higher efficiency and factor substitution in end uses and of decarbonization of the fuel mix. Fossils fuels without carbon capture and sequestration (CCS) are progressively phased out by nuclear, renewables, and fossil fuel power plants with CCS. When the carbon tax is high, bioenergy with CCS plays an important role. Coal power plants without CCS disappear by the end of the century in both the RCP4.5 and the RCP2.6 scenarios. The RCP2.6 scenario is so stringent that coal with CCS is also progressively phased-out because the penalty on the uncaptured emissions is very expensive. Thus climate policy also delivers local environmental benefits by greatly reducing harmful emissions released during coal combustion. Figure 4 illustrates how the power generation mix evolves over the century in the Reference and in the climate policy scenarios.



Figure 4. The electricity generation mix at global level (left column) and in the EU (right column).



Notes: costs discounted using the endogenous interest rate of the model.

Figure 5. Consumption and GDP losses in the tax scenarios.

We note that in the limited technology scenarios we still achieve the same target for overall radiative forcing and hence for carbon concentrations of the full flexibility scenarios. The only differences between the full flexibility scenario and scenario with limited adaptation are the technology mix and the cost of the policy. Emissions from the power sector are also different, as we document below.

All the mitigation measures implemented in the tax scenarios are costly because they reduce the overall efficiency of the economy. As a result economic growth is slower in the tax scenarios and aggregated consumption and GDP decline with respect to the Reference scenario. Figure 5 shows the aggregated consumption and GDP losses from 2010 to 2100, evaluated using the endogenous interest rate calculated in WITCH. These costs are not net of the economic benefit of reduced global warming and of reduced local pollution.

Consumption and GDP losses at global level range from almost zero in the RCP6.0 scenario to about 5% and 6% in the RCP2.6 scenario, respectively. In Europe the cost of the RCP2.6 scenario is lower than at global level, and it is equal to about 2% in terms of consumption and about 3% in terms of GDP.

5.2 Impact on emissions

Each climate mitigation scenario results in a remarkable decrease in the total volume of polluting emissions, with the exception of $PM_{2.5}$ and nickel emissions (Table 2). The increase in these two pollutants is a consequence of substituting fossil fuels with biomass, for which the EMEP/EEA inventory reports the highest emission factors among all fuels. As biomass is carbon neutral, it becomes a viable alternative to fossil fuels in all mitigation scenarios. In particular, the RCP2.6 scenario has a very large use of biomass with carbon capture and sequestration to compensate for emissions in other sectors. As a consequence, the RCP2.6 scenario displays the largest increase in $PM_{2.5}$ pollutants.

In the RCP4.5 scenarios with limited technological adaptation, emissions of air pollutants change, although not dramatically. The only exception is $PM_{2.5}$ when supply and trade of biomass is limited (RCP4.5-BIO). The fully flexible scenario (RCP4.5) leads to an increase of $PM_{2.5}$ emissions equal to 3,056 kt (with respect to the Reference scenario), while the constrained scenario leads to a reduction of 9 kt of $PM_{2.5}$ (with respect to the constrained Reference scenario). Annual changes in emissions of $PM_{2.5}$ and SO_2 for each climate mitigation scenarios are displayed in Figure 6.

Scenario	NMVOC (kt)	NOx (Mt)	PM2.5 (kt)	SOx (Mt)	Cd (t)	As (t)	Ni (t)	Pb (t)	Hg (t)	Cr (t)	CO ₂ (Gt)
RCP2.6	-724	-182	13,686	-640	-494	-4,376	-7,904	-3,078	-923	-2,362	-120
RCP4.5	-574	-91	3,056	-306	-268	-2,385	3,775	-2,083	-504	-1,403	-74
RCP6.0	-345	-33	-144	-97	-94	-861	4,516	-864	-186	-538	-17
RCP4.5-EFFIC	-365	-86	3,890	-314	-273	-2,370	253	-1,997	-492	-1,376	-80
RCP4.5-REN	-323	-72	2,989	-261	-232	-1,981	-837	-1,687	-409	-1,154	-65
RCP4.5-BIO	-481	-71	-9	-247	-257	-2,078	-1,018	-2,037	-420	-1,285	-65

Table 2. Cumulative difference in emission volumes in Europe for each climate mitigation scenario compared to the corresponding reference scenario for the period 2015 – 2100.



Figure 6. Annual change in emission volumes of PM2.5 (left) and SO2 (right) for Europe for each climate mitigation scenario.

5.3 Ancillary benefits in total and per ton of abated CO₂

The ancillary benefits of climate mitigation are displayed in Table 3. The first two columns display total CO_2 emissions (CO_2) and total local pollution costs (PC). The third and the fourth column present the change of CO_2 emissions (ΔCO_2) and the ancillary benefits (AB= Δ PC). The fifth column displays ancillary benefits per reduced ton of CO_2 (AB/ ΔCO_2). All results are relevant to Europe only and all nominal values are expressed in present value of 2005 Euro, discounted using the endogenous interest rate from the WITCH model.

In the RCP2.6 scenario the level of CO_2 -eq concentrations is kept at 490 ppm thanks to a massive reduction of GHG emissions compared to the Reference scenario (120 Gt over the period 2015-2100). Due to this sharp reduction of GHG emissions also local pollution collapses and the discounted cost of pollution over the century drops from ξ 3,394B to ξ 816B (76% reduction). The ancillary benefits are higher than ξ 2,500B, which implies an average benefit of about ξ 21.6 per abated ton of CO_2 -eq. in Europe. The RCP4.5 generates the second largest ancillary benefits, amounting to more than ξ 1,061B. These benefits result from 31% cumulative reduction in the pollution cost in Europe. CO_2 emissions are reduced by 73 Gt in the RCP4.5 scenario, which implies 14.4 ξ /tCO₂-eq of local benefits per abated. The RCP6.0 – with only 15% CO₂-eq reduction and 9% reduction of the pollution costs – leads to one order of magnitude smaller cumulative ancillary benefits than the RCP2.6. The magnitude of the ancillary benefits per ton of CO_2 -eq avoided is, however, very similar. This result is explained by the strong similarity of the power generation mix across the three scenarios. The share of total emissions reductions from the power sector is also rather constant across scenarios. The RCP4.5 scenarios with limited adaptation have ancillary benefits that are similar to those found for

the unconstrained RCP4.5 scenario. Our results are in line with other studies (see Section 2), but our study covers a longer time horizon, it considers more abatement options and more climate policy scenarios.

	(1)	(2)	(3)	(4)	(5)
	CO ₂	PC	ΔCO_2	AB	$AB/\Delta CO_2$
	Mt	€bn.	Mt	€bn.	€
Full technological flexibility					
Reference scenario					
SSP2	118,396	3,394			
Climate policy scenarios					
RCP2.6	-1,144	816	-119,540	-2,578	21.56
RCP4.5	44,465	2,333	-73,931	-1,061	14.35
RCP6.0	101,807	3,090	-16,589	-304	18.33
Limited technologies					
Reference scenario					
SSP2-EFFIC	118,361	3,545			
SSP2-REN	109,371	3,321			
SSP2-BIO	108,453	3,299			
Climate policy scenario					
RCP4.5-EFFIC	38,614	2,397	-79,746	-1,148	14.40
RCP4.5-REN	44,532	2,352	-64,839	-969	14.94
RCP4.5-BIO	43,299	2,345	-65,155	-954	14.65

Notes: Cumulative and discounted over the period 2015-2100 (Euro 2005).

Table 3. Present value of ancillary benefits in Europe for climate mitigation scenarios.

Table 3 displays discounted ancillary benefits. Without discounting, climate mitigation yields in Europe, on average, ancillary benefits in the range of ≤ 36 (RCP4.5) to ≤ 50 (RCP6.0) per abated ton of CO₂eq for the whole period until 2100. Ancillary benefits of the stricter mitigation (RCP2.6) are about 44 \leq /tCO₂eq during 2015-2050 and after 2080 the benefits are over ≤ 50 per ton. Benefits for the RCP4.5 scenario follow an inverted U-shape form; the benefits are about 30 \leq per ton until 2025, then they are rising at ≤ 60 level around 2040-2044, and then they again go down reaching ≤ 27 per ton CO₂eq in the 2080's. Under mild climate mitigation (RCP6.0), the ancillary benefits start at around ≤ 30 , then they reach ≤ 50 . In absolute terms, total benefits, as cumulated over the period 2015-2100, are slightly over $\leq 5,600B$ (RCP2.6), $\leq 2,600B$ in RCP4.5, and only about $\leq 800B$ in RCP6.0.

We then compare ancillary benefits (AB) and the total cost of electricity generation (TC), both discounted and in present values. Table 4 displays both benefits and costs as cumulated over the period that always starts from 2015 and ends in the year at the end of given period. Both are derived as a difference with respect to the corresponding Reference scenario. For the whole period 2015-2100, the ancillary benefits exceed the additional cost of electricity generation in the RCP2.6 as well as in the RCP4.5; the RCP6.0 results in cost savings.

	RCP2.6		RCP4.5		RCI	P6.0	RCP4.5	-EFFIC	RCP4.	5-REN	RCP4.5	-BIOM
	AB	тс	AB	тс	AB	тс	AB	тс	AB	тс	AB	тс
2019	237	-59	8	19	7	18	1	2	1	2	1	2
2024	440	-70	20	67	17	51	5	34	4	27	4	28
2029	637	-20	81	128	69	77	28	82	24	68	25	70
2034	833	33	187	177	142	96	77	140	68	106	69	111
2039	1,009	104	291	232	207	104	145	191	115	160	119	172
2044	1,171	180	393	277	262	105	218	181	173	214	184	219
2049	1,316	268	489	258	301	95	290	204	240	215	250	222
2059	1,584	447	652	307	357	68	430	223	369	309	378	248
2069	1,830	660	773	259	378	18	568	305	491	335	502	302
2079	2,070	882	880	207	396	-71	702	431	605	403	619	374
2089	2,308	1,119	970	144	373	-205	884	624	757	522	761	459
2099	2,552	1,349	1,052	62	311	-360	1,121	786	947	683	935	547
2100	2,578	1,375	1,061	54	304	-379	1,148	817	969	703	954	564
ratio AB/TC	1.87		19.	56	-0	-0.80		11	1.38		1.69	

Note: Present value of cumulative benefits and costs up to the year indicated in the first column. AB – ancillary benefits, TC – total costs of electricity generation. All monetary values in Billion Euro 2005.

Table 4. Present value of cumulative ancillary benefits and total costs of electricity generation in Europe,2015-2100.

Table 5 provides a comparison between the undiscounted (real) value of ancillary benefits and the carbon price per ton of CO_2 for each climate mitigation scenario and their progress until 2100. RCP6.0 implies the lowest carbon price, from $\notin 2$ to $\notin 12$ per ton of CO_2 eq. We can also find that, on average for the whole period, ancillary benefits per ton of CO_2 eq are 7-times larger than the carbon price. Due to higher carbon prices in stricter climate mitigation scenarios, this ratio is only 0.5 in RCP4.5 and 0.06 in RCP2.6. Our results are however still in line with other studies; for instance Grossman et al. (2011) found the ratio of ancillary benefits on permit price in a range between 40% and 250% in 2015 (with CO_2 price at \$29), and between 18% and 125% in 2030 (\$61).

		RCP2.6 R(RCP4.5			RCP6.0		R	CP4.5-EFF	IC		RCP4.5-RE	N	RCP4.5-BIOM		
	AB	AB/CO ₂	p/CO ₂	AB	AB/CO ₂	p/CO ₂	AB	AB/CO ₂	p/CO ₂	AB	AB/CO ₂	p/CO ₂	AB	AB/CO ₂	p/CO ₂	AB	AB/CO ₂	p/CO ₂
	bn.€	€	€	bn.€	€	€	bn.€	€	€	bn.€	€	€	bn.€	€	€	bn.€	€	€
2015-19	291	48	104	7	33	11	6	32	2	2	44	15	1	50	11	1	50	12
2020-24	568	44	129	22	29	13	18	26	3	7	45	18	6	45	13	6	45	14
2025-29	846	43	159	95	47	16	77	51	3	43	51	22	37	52	16	37	51	18
2030-34	1,140	44	194	243	60	20	170	54	3	122	69	27	108	72	20	109	71	21
2035-39	1,434	43	238	409	59	24	270	53	4	241	72	33	191	70	24	197	70	26
2040-44	1,733	43	292	593	60	30	364	53	4	377	60	40	299	71	30	318	72	32
2045-49	2,032	43	357	792	57	36	449	51	4	521	40	49	435	64	37	453	60	39
2050-59	2,644	44	436	1,171	43	44	591	49	5	827	34	59	718	44	44	734	42	48
2060-69	3,300	46	655	1,522	32	65	698	45	6	1,167	30	86	1,016	32	65	1,036	29	70
2070-79	4,010	48	982	1,872	28	95	785	43	7	1,559	29	125	1,354	29	96	1,373	28	103
2080-89	4,756	51	1,460	2,219	27	138	829	49	9	2,051	31	182	1,757	30	138	1,758	29	151
2090-99	5,555	55	2,128	2,598	28	198	829	-1*	10	2,674	34	261	2,259	32	199	2,202	31	218
2100	5,639	58	2,951	2,638	30	281	825	25*	12	2,749	36	370	2,317	34	283	2,255	33	311
average AB/CO ₂		47.17			35.69			49.75			34.47			35.73			34.61	
ratio AB/price		0.06			0.52			7.11			0.41			0.59			0.53	

Note: $AB - cumulative ancillary benefits over the period 2015 until the end of the year shown on the respective line (for instance, 2030-39 indicates the benefits over 2015 to 2039). <math>AB/CO_2 - average$ ancillary benefits per ton reduced CO_2 in given period; $p/CO_2 - carbon$ price per ton CO_2 ; all values in real undiscounted 2005 Euro. The indicators related to CO_2 emissions are expressing average annual value for given period shown in the first column. * Starting from 2090, CO_2 emissions are slightly higher than in the SSP2 reference scenario. In 2100, RCP6.0 generates ancillary damage, instead of benefits, but they are negligible in size.

Table 5. Real values of (cumulative) ancillary benefits, average annual ancillary benefits and carbon price per ton of CO₂eq abated, 2015-2100 (values in Euro 2005).

5.4 Ancillary benefits for European countries: winners and losers

Lastly, we are interested in understanding the distribution of the ancillary benefits of climate change mitigation policy across European countries. Unfortunately we do not have country-by-country energy scenarios. WITCH separates Europe in two large blocs: Western and Eastern Europe. We build energy scenarios for 28 European countries assuming that each country's share of the total regional energy use remains constant over time and under alternative climate mitigation policies. Following equation 1, we compute pollution costs for each of the 28 countries that are born by each and every other European country and by people living outside of the 28 European countries. Having 28 x 29 pollution cost values, we can then derive total pollution costs that are associated with impacts on emission receptors in (1) country *i* due to emissions released by the same country, (2) other European countries *j* (*j*≠*i*) and 3) the rest of Europe that are due to emissions released by country *i*. It is then straightforward to derive the pollution costs – both due to domestic (1) or imported pollution (2) – for any country in Europe. Burden exported from Europe is measured by (3).

The total value of ancillary benefits for the RCP2.6 scenario between 2015 and 2100 is estimated to be equal to $\leq 2,578B$ and about 73% of these benefits have a domestic origin, that is, the local benefits are enjoyed by residents of the European country in which the abatement in local air ambient pollution occurs. The remaining share of ancillary benefits is divided between beneficiaries from European countries ($\leq 570B$, about 22%) and beneficiaries from rest of the world ($\leq 134B$, about 5%).

Switching to a country-level analysis, we find that (1) each of the 28 European countries is better off in terms of air quality improvement if RCP2.6 mitigation policy is implemented. In absolute terms, European countries benefit between 2015 and 2100 from a modest 88m \in (in Cyprus) to 1,192B \in (in Germany). (2) Countries' shares of abatement effort and share of the benefits that are enjoyed by their residents vary significantly, as shown in Figure 7. Germany generates the largest share of the ancillary benefits, about 44%, followed by the United Kingdom with a contribution of 17%, Italy and Poland both with 6%. These countries also enjoy the largest shares of local benefits. Half of the 28 European countries do not contribute more that 5% to total ancillary benefits, but also do not enjoy more than 5% of the total ancillary benefits. (3) In three European countries – Switzerland, Lithuania and Sweden – the RCP2.6 mitigation scenario results in higher non-GHG emissions than in the Reference case and as a result the generated ancillary benefits are negative in these countries, i.e. RCP2.6 generates damage; thanks to reduced import of emission air quality would be overall also improved.



Notes: RCP2.6 scenario. Percentage of EU reduction of costs. Figure 7. Countries' contribution to total ancillary benefits and to total effort of their generation in Europe.

From Figure 8 it is also possible to infer which countries are net externality producers and which countries are instead net beneficiaries. We define net producers of the ancillary benefits those countries whose share of total benefits generated in Europe is larger than their share of total benefits that are enjoyed by its residents. For example, Bulgaria's contribution to cumulative ancillary benefits due to its non-GHG emission reductions is 0.8% of the total benefits in Europe (€20.7B), while Bulgarians receive only 0.1% of total ancillary benefits in Europe (€2.5B). This implies that Bulgaria contributes more to overall ancillary benefits in Europe than it receives in the RCP2.6 scenario. Such a country can be considered to relatively lose out as a result of the GHG mitigation policy, although we highlight that each country is better off with respect to overall pollution costs in RCP4.5. Estonia, Ireland and Slovenia face the same situation as Bulgaria. Cyprus, Portugal, Finland, Romania, Czech Republic, Greece, Poland and Denmark belong among the net producers as well, but their relative contribution to the overall ancillary benefits is not as large compared to their share of benefits that their residents would enjoy as it is in the first group of net producing countries. The contribution of large countries, such as Germany, France, Italy, Spain, or the United Kingdom, but also of the Netherlands, Slovakia and Hungary is roughly equivalent to their benefits. Belgium, Austria, Latvia, Norway and Luxemburg are the relative winners in Europe since they all receive more ancillary benefits than they generate.



Figure 8. Contribution to ancillary benefits generated relative to the share of benefits that a country would enjoy in RCP2.6, 2015-2100.

The left panel of Figure 9 displays the percentage of ancillary benefits that result from emission reduction in one country which are 'exported' elsewhere. The rest of the benefits have a domestic origin, i.e. they happen due to emission abatement in the same country. The right panel shows 'imported' ancillary benefits as a fraction of total ancillary benefits. These benefits are enjoyed thanks to the emission reductions in other countries. For instance, emission reductions in Italy will mainly benefit the local population and local assets. Only 11% of ancillary benefits are 'exported' to other countries. Cyprus, Italy, Spain or Greece do not benefit much from emission reductions in other countries. Most of the benefits in Luxembourg, Latvia, Norway or Belgium are instead due to the external emissions reductions. In the case of Sweden, Lithuania and Switzerland the share exceeds 100% which describes a situation when 'imported' benefits exceed total benefits. This happens when ancillary benefits due to domestic abatement are negative, that is when the RCP2.6 climate mitigation policy increases local pollution.



Figure 9. Exported (left panel) and imported (right panel) ancillary benefits from pollution reduction.

6 Conclusions and remarks

This paper presents estimates of the air quality ancillary benefits of GHG emission reductions. We consider three climate policy scenarios that achieve three standard radiative forcing levels in 2100: 6.0, 4.5 and 2.6 W/m². These three levels correspond to concentrations of GHG equal to 850, 650, and 490 ppm CO_2 equivalent, which result in global mean temperature increases of 3.7°C, 3.0°C and 2.0°C, respectively. These radiative forcing targets are achieved by introducing a uniform tax on all GHG emissions in the Integrated Assessment Model WITCH, assuming full as well as limited technological flexibility. Our scenarios cover the whole world for the period 2015-2100, however, we assess only ancillary benefits from emissions reductions in electricity generation in Europe.

As emitting GHG becomes more expensive, WITCH projects investment in carbon-free technologies for the energy sector, energy efficiency and R&D to develop carbon-free backstop technologies, substitutes the energy input with other factors and invests in a series of activities to reduce CO_2 emissions from deforestation and other non- CO_2 gases. As a result economic growth is slower in the tax scenarios. In Europe, mitigation policies are usually less costly than in other regions of the world, with cumulated consumption and GDP losses amounting to about 2% and 3% in the RCP2.6 scenario, respectively, if a discount rate based on the endogenous interest rate of WITCH is applied. Compared to the Reference scenario, until 2100 cumulative GHG emissions are reduced by 14% in the RCP6.0, but they are 62% smaller in the RCP4.5 and RCP2.6 leads even to negative cumulative GHG emissions due to the use of biomass with carbon capture and sequestration.

The resulting consumption patterns of fossil fuels in the European electricity sector are used to estimate the physical and economic benefits of pollution reductions on human health and on key assets by implementing the most advanced version of the ExternE methodology with its Impact Pathway Analysis. The ancillary benefits are derived as the difference between pollution costs for the mitigation and the reference scenario, when impacts on human health, crop yield, materials, and biodiversity associated with emissions of PM2.5, PM10, SO₂, NOx, and NMVOC are considered.

In sum, our estimates of the ancillary benefits of reducing non-GHG polluting emissions in the electricity sector in Europe are in line with estimates in the literature and very close to $33 \in$ per ton of CO₂ abated, the mean value of the benefits reported in the review by Nemet et al. (2010) for developed countries. Specifically, in the RCP2.6 mitigation scenario, we quantify these benefits in Europe at more than $\leq 2.5T$ during 2015-2100, which implies average ancillary benefits of about 21.6 \leq /t CO₂eq abated in the European electricity sector. Less strict scenarios generate overall smaller local benefits, but the magnitude of ancillary benefits per ton of CO₂eq abated is only slightly smaller (≤ 14.4 , or ≤ 18.3 , respectively) than under stricter mitigation. In real terms, without discounting, the ancillary benefits are in the range of ≤ 36 to ≤ 50 per ton of CO₂eq abated and the unit value of the benefits for most mitigation scenarios slightly increases over time.

On average, ancillary benefits per ton of CO_2 eq abated in Europe are about 7-times larger than the projected carbon price by WITCH for the RCP6.0 mitigation scenario. As the carbon price increases in the RCP4.5, this ratio declines to 0.5, and it gets as low as 0.06 in the RCP2.6. A lower ratio for larger carbon prices has been frequently documented elsewhere (Grossman et al. 2011) and our results are in fact in line with general empirical evidence. For the whole period, cumulative ancillary benefits also exceed cumulative additional cost of electricity generation in Europe, with an exception in the RCP6.0 that actually results in overall cost savings in the power sector.

At country level we find that about 73% of total ancillary benefits attributable to the RCP2.6 mitigation scenario have a domestic origin, while the rest of the benefits are enjoyed in other countries than in the one where the damaging emissions were abated. There are several key findings worth mentioning: first, each of the 28 European countries is better off if RCP2.6 mitigation policy is implemented, although the absolute value of the benefits varies significantly across countries, reflecting their size, primary energy use, and fuel and technology mix to generate electricity. Second, Germany is the key player; it contributes to ancillary benefits by about 44% and receives approximately the same share of total ancillary benefits generated in Europe. Other large European countries such as the United Kingdom, Italy and Poland also considerably contribute to total ancillary benefits. Many other European countries are either small or have cleaner energy mixes and do not contribute much to European-wide ancillary benefits. 14 out of 28 countries generate (and enjoy) less than 5% of total European ancillary benefits. Third, there are relative winners and losers with respect to ancillary benefits in Europe. Large European countries are neutral - they contribute to ancillary benefits as much as they benefit. Hungary, Belgium, Austria, Latvia and Norway benefit more than they contribute to ancillary benefits. Although Switzerland, Lithuania and Sweden generate domestically ancillary damage rather than benefit, overall they win as well due to larger ancillary benefits imported from other European countries.

Technologically constrained RCP4.5 scenarios lead to a higher price of carbon, but the price effect on all emissions is relatively small compared to the emission predictions for the scenario with full flexibility. Overall, limited flexibility in energy efficiency, biomass supply and renewable energy technologies does not yield substantially different results from the ones that we obtain for the mitigation scenario with full flexibility.

The literature suggests that the ancillary benefits of climate mitigation can be very large and are thus policy relevant. Our results confirm previous studies. As pointed out by Burtraw et al. (2003), not considering ancillary benefits could lead to an incorrect assessment of the net costs of mitigation policies and to an incorrect identification of 'no regrets' levels of GHG mitigation. For instance, Nam et al. (2013), by using the EPPA5 model, found that if China achieves its SO₂ and NOx emission reduction targets, as proposed in its 12th Five Year Plan, the corresponding carbon-mitigation potential exceeds China's official 17% CO₂ intensity reduction goal. Hence, if these ancillary benefits can be measured in monetary terms, they should be included in the cost calculation of climate policy (Davis et al. 2000). However some caveats apply.

First, in an ideal world each externality should be addressed using a specific tool. For example, most of the externalities from fossil fuels combustion are additive and should be corrected using taxes on fuel use that reflect the marginal cost of pollution (Heine et al. 2012).²¹ Efficient regulation thus requires the imposition of a penalty on fuel use that is equal to the sum of the marginal cost of each pollutant. Climate policy cannot substitute local pollution policies and *vice versa*. Unfortunately, this efficient solution is rarely implemented. If local pollution effects are not internalized, a carbon tax calibrated to reflect the marginal cost of carbon emissions has positive spillovers. If climate policy is not implemented taxes aimed at reducing local pollution may have a positive global spillover if the fuel mix shifts towards cleaner fuels. If local pollution is reduced using end-of-pipe technologies carbon emissions may remain constant or may increase due to efficiency losses This externality should be included in the cost calculations of local pollution policies.

Second, the existence of ancillary benefits should not be a reason for increasing the carbon tax or to make climate regulation more stringent. The goal of carbon taxes is to reduce the climate externality and the carbon price should be set to reflect the social cost of carbon. If climate policy also improves air quality, then air quality policy can be less stringent than it would be if climate policy was not implemented. This benefit of GHG mitigation should not be neglected but the social cost of carbon does not change.

Finally, we appreciate Ian Parry's et al. (2014) suggestion to "understand how much carbon emissions reduction is in the self-interest of countries" and to derive nationally efficient carbon prices. We are however more cautious than to call for "an approach that builds on national self-interest and spurs a race to the top in low-carbon energy solutions" (Parry 2014.). It is true that policies regulating local damage are more acceptable by the public than policies which have local costs but global benefits far in the future. The focus on local benefits may increase the political acceptability of carbon taxes. However, it remains unclear what would happen if the carbon tax had to increase above the level that is deemed optimal to reduce local pollution. We find that the +2°C compatible scenario has a carbon price that is always much higher than what would be justified by local benefits alone. Although we do not account for all possible ancillary benefits, it seems plausible to assume that pursuing local benefits alone will not keep the increase of global mean temperature below +2°C.

There are several limitations in our modelling approach that may require further research.

²¹ In some cases climate and local pollution have a multiplicative effect. For example ozone formation depends on the joint combination of local pollutants and particular climatic conditions. However, these are special cases that require special treatment. It is safe to assume that short-term climatic conditions are not affected by the carbon tax and that long-term ozone formation is not affected by the present level concentrations of local pollutants. The same reasoning applies to environmental regulation that aims to curb local pollutants.

First, since the quantification of ancillary benefits is based on a soft-link of WITCH global model and ExternE's Impact Pathway Approach, there is no optimal joint management of local and global externalities. Within the very recent LIMITS project, air quality model has been made endogenous in WITCH model, which will allow later analysing the joint management of the two externalities.

Second, we use WITCH to determine the economic and technological impacts at global level, while the non-GHG impacts and ancillary benefits are quantified for Europe and only from non-GHG emission abatement in electricity sector. Our assessment is hence not complete.

Third, we are interested in finding out the distribution across European countries of the pollution costs and of the ancillary benefits of climate change mitigation policy. The WITCH model provides energy scenarios for two large blocs in Europe: Western and Eastern Europe. We therefore build energy scenarios for 28 European countries assuming that each country's share of total energy use in the European bloc remains constant over time and under alternative climate mitigation policies. In reality, fuel-mix may change differently across countries in the bloc. Assuming that each country will *mimic* others within the bloc is quite a reasonable assumption, especially in the long run.

Fourth, ancillary benefits are determined by emission-fuel coefficients that are derived from the EMEP EEA air pollutant emission inventory guidebook (EMEP/EEA 2013). More sources may provide a wide range of the emission-fuel coefficients that might be used in a sensitivity analysis to estimate uncertainty in our estimate of ancillary benefits. The emission factors of fuels also remain constant over the entire period that actually implies no improvement in the efficacy of the end-of-pipe non-GHG abatement technologies. Incorporating dynamic improvement in the abatement efficacy would significantly improve our calculations.

We leave all these possible extensions to future research.

Acknowledgement

This research received funding from the European Union's Seventh Framework Programme (FP7/2007–2013) under the grant agreement n° 266992 GLOBAL-IQ "Impacts Quantification of Global changes". Preparation of the manuscript received a support from project ECOCEP (Economic Modelling for Climate-Energy Policy) funded by the People Programme (Marie Curie Actions) of the European Union's Seventh Framework Programme FP7-PEOPLE-2013-IRSES, Grant Agreement No 609642. This support is gratefully acknowledged. The authors would like to thank Laura Henderson Macháčková for proofreading this article. Responsibility for any errors remains with the authors.

References

Aaheim HA, Aunan K, Seip HM (1997) Social benefits of energy conservation in Hungary: An examination of alternative methods of evaluation. CICERO Center for International Climate and Environmental Research – Oslo

Abt Associates and Pechan-Avanti Group (1999) Co-Control Benefits of Greenhouse Gas Control Policies. Prepared for Office of Policy, U.S. Environmental Protection Agency, Contract No. 68-

W4-0029, http://www.epa.gov

Acemoglu D, Aghion P, Bursztyn L, Hemous D (2009) The Environment and Directed Technical Change. Am Econ Rev 102(1): 131-166

Alcamo J, Mayerhofer P, Guardans R, et al (2002) An integrated assessment of regional air pollution and climate change in Europe: findings of the AIR-CLIM Project. Environ Sci Policy 5:257–272. doi: 10.1016/S1462-9011(02)00037-0

Aunan K, Aaheim HA, Seip HM (2000) Reduced damage to health and environment from energy saving in Hungary. Ancillary Benefits and Costs of Greenhouse Gas Mitigation 397–411

Barker T, Rosendahl KE (2000) Ancillary Benefits of GHG Mitigation in Europe: SO2, NOx and PM10 reductions from policies to meet Kyoto targets using the E3ME model and Externe valuations. In: OECD Ancillary Benefits and Costs of Greenhouse Gas Mitigation, Proceedings of an Expert Workshop, Paris 413–450

Barrett S, Stavins R (2003) Increasing Participation and Compliance in International Climate Change Agreements. Int Environ Agreem-P 3:349–376. doi: 10.1023/B:INEA.0000005767.67689.28

Bauer N., Bowen A, Brunner S, Edenhofer O, Flachsland C, Jakob M, Stern N (2009) Towards a GlobalGreen Recovery. Recommendations for Immediate G20 Action. Report submitted to the G20 London Summit, 2 April 2009.

Bell ML, Davis DL, Cifuentes LA, et al (2008) Ancillary human health benefits of improved air quality resulting from climate change mitigation. Environ Health 7:41.

Bollen J, van der Zwaan B, Brink C, Eerens H (2009) Local air pollution and global climate change: A combined cost-benefit analysis. Resour Energy Econ 31(3):161–181. doi:

10.1016/j.reseneeco.2009.03.001

Bosetti V, Carraro C, Galeotti M, Massetti E and Tavoni M (2006) WITCH - A World Induced Technical Change Hybrid Model. Energ J 27(Special Issue 2):13-38.

Bosetti V, Tavoni M, De Cian E, Sgobbi A (2009) The 2008 WITCH model: New model features and baseline. Nota di lavoro // Fondazione Eni Enrico Mattei: Sustainable development

Bosetti V, Massetti E, Tavoni M (2007) The WITCH Model. Structure, Baseline, Solutions. FEEM Nota di Lavoro No. 10.2007

Bovenberg AL (1999) Green tax reforms and the double dividend: an updated reader's guide. Int Tax Public Finan 6(3):421–443

Boyd R, Krutilla K, Viscusi WK (1995) Energy Taxation as a Policy Instrument to Reduce CO₂ Emissions: A Net BenefitAnalysis. J Environ Econ Manag 29(1): 1-25.

Brendemoen A, Vennemo H (1994) A Climate Treaty and the Norwegian Economy: A CGE Assessment. Energ J 15(1): 77-91

Burtraw D, Krupnick A, Palmer K, Paul A, Toman M, Bloyd C (1999) Side effects of Reduced Air Pollution in the US from Moderate Greenhouse Gas Mitigation Policies in the Electricity Sector, Resources for the Future Discussion Paper 99-51 (REV), Resources for the Future, Washington DC

Burtraw D, Krupnick A, Palmer K, et al (2003) Ancillary benefits of reduced air pollution in the US from moderate greenhouse gas mitigation policies in the electricity sector. J Environ Econ Manag 45:650–673. doi: 10.1016/S0095-0696(02)00022-0

Burtraw D, Linn J, Palmer K, Paul A (2014) The Costs and Consequences of Clean Air Act Regulation of CO_2 from Power Plants. Am Econ Rev: Papers & Proceedings 104(5): 557–562

Carraro C, Galeotti M, Gallo M (1996) Environmental taxation and unemployment: Some evidence on the "double dividend hypothesis" in Europe. J Public Econ 62(1):141–181

Cifuentes LA, Sauma E, Jorquera H, Soto F (2000) Preliminary Estimation of the Potential Ancillary Benefits for Chile. In: OECD Ancillary Benefits and Costs of Greenhouse Gas Mitigation, Proceedings of an Expert Workshop, Paris 237-262

Czajkowski M, Ščasný M (2010) Study on benefit transfer in an international setting. How to improve welfare estimates in the case of the countries' income heterogeneity? Ecol Econ 69(12):2409–2416. doi: 10.1016/j.ecolecon.2010.07.008

Davis DL, Krupnick A, McGlynn G (2000) Ancillary benefits and costs of greenhouse gas mitigation. In: OECD Ancillary benefits and costs of greenhouse gas mitigation, Proceedings of an Expert Workshop, Paris 9-49

Desaigues B, Ami D, Bartczak A, et al (2011) Economic valuation of air pollution mortality: A 9country contingent valuation survey of value of a life year (VOLY). Ecol Indic 11:902–910. doi: 10.1016/j.ecolind.2010.12.006

Dessus S, O'Connor D (1999) Climate Policy Without Tears. Organisation for Economic Co-operation and Development, Paris

Dudek D, Golub A, Strukova E (2003) Ancillary Benefits of Reducing Greenhouse Gas Emissions in Transitional Economies. World Dev 31(10):1759–1769. doi: 10.1016/S0305-750X(03)00139-6

Ebi KL, Hallegatte S, Kram T, et al (2014) A new scenario framework for climate change research: background, process, and future directions. Climatic Change 122:363–372. doi: 10.1007/s10584-013-0912-3

EC (2005) ExternE - Externalities of Energy. Methodology 2005 Update, Luxembourg: Office for Official Publications of the European Communities.

EMEP/EEA (2013) 1.A.1 Combustion in energy and transformation industries. EMEP/EEA air pollutant emission inventory guidebook - 2013 — European Environment Agency (EEA).

ESPREME 2007. Integrated Assessment Of Heavy Metal Releases In Europe, Exposure Response Functions, Searl, A. "Workpackage 6 – D05a, Exposure response functions for HM impacts on human health", 2004. http://espreme.ier.uni-stuttgart.de/

European Commission (1995). Externalities of Energy: Volume 1: Summary; Volume 2: Methodology; Volume 3: Coal and Lignite; Volume 4: Oil and Gas; Volume 5: Nuclear; Volume 6: Wind and Hydro Fuel Cycles. European Commission, Directorate-General XII. Science, Research and Development.

Fantke P (2008) Parameterisation of the environmental fate and exposure assessment of WATSON.

Garbaccio RF, Ho MS, Jorgenson DW (2000) The Health Benefits of Controlling Carbon Emissions in China. In: OECD Ancillary Benefits and Costs of Greenhouse Gas Mitigation, Proceedings of an Expert Workshop, Paris 343-376

Glomsrød S, Vennemo H, Johnsen T (1992) Stabilization of Emissions of CO2: A Computable General Equilibrium Assessment. Scand J Econ 94:53–69. doi: 10.2307/3440468

Goulder LH (1995) Environmental taxation and the double dividend: A reader's guide. Int Tax Public Finan 2(2):157–183. doi: 10.1007/BF00877495

Groosman B, Muller NZ, O'Neill-Toy E (2011) The ancillary benefits from climate policy in the United States. Environ Resource Econ 50(4):585–603

Hammitt JK (2007) Valuing changes in mortality risk: lives saved versus life years saved. Rev Environ Econ Policy 1(2):228–240

Heine D, Norregaard J, Parry I (2012) Environmental tax reform: principles from theory and practice to date.

Hohmeyer O (1988) Social costs of energy consumption. a Report prepared under contract for the Commission of the European Communities, Berlin: Springer Verlag

Holland M, Amann M, Heyes C, et al (2011) The reduction in air quality impacts and associated economic benefits of mitigation policy: summary of results from the EC RTD climate cost project. Technical Policy Briefing Note Series. Oxford: Stockholm Environment Institute.

Holland MR, Forster D, King K (1999) Cost-benefit analysis for the protocol to abate acidification, eutrophication and ground level ozone in Europe. Ministerie van Volkshuisvesting, Ruimtelijke Ordening en Milieubeheer, Centrale Directie Communicatie

Intergovernmental Panel on Climate Change (IPCC) (2001) Climate change 2001: mitigation. Contribution of Working Group III to the Third Assessment Report of the IPCC. In: Metz B, Davidson O, Swart R, Pan J (Eds.). Cambridge, MA: Cambridge University Press.

IVM (1999) Economic evaluation of air quality targets for fine particles, SO2, NO2 and lead. European Commission DG XI, Brussels

Kiuila O, Markandya A, Ščasny M, et al (forthcoming) The Economic and Environmental Effects of Taxing Air Pollutants and CO2: CGE Model With Abatement Technologies And Sector-Specific Emission Coefficients. Under review in Economic System Research

Leimbach M, Massetti E, Bosello F, et al (2014) Global challenges with limited adaptation, GLOBAL IQ Deliverable 4.1

Máca V, Melichar J, Ščasný M (2012) Internalization of External Costs of Energy Generation in Central and Eastern European Countries. J Environment Dev 21(2):181–197. doi: 10.1177/1070496512442504

Markandya A, Armstrong BG, Hales S, et al (2009) Public health benefits of strategies to reduce greenhouse-gas emissions: low-carbon electricity generation. Lancet 374:2006–2015. doi: 10.1016/S0140-6736(09)61715-3

Massetti E, Bosello F, Carrara S, et al (2014) Scenarios of global challenges with full adaptation, GLOBAL-IQ Deliverable 5.1

Mendelsohn RO (1980) An economic analysis of air pollution from coal-fired power plants. J Environ Econ Manag 7(1):30–43. doi: 10.1016/0095-0696(80)90020-0

Meyer B (1998) Modellierung der Nachhaltigkeitslücke: eine umweltökonometrische Analyse; mit 38 Tabellen. Physica-Verlag

Morgenstern RD (2000) Baseline issues in the estimation of the ancillary benefits of greenhouse gas mitigation policies. In: OECD Ancillary Benefits and Costs of Greenhouse Gas Mitigation 95–122

Muller NZ, Mendelsohn RO (2007) Measuring the damages of air pollution in the United States. J Environ Econ Manag 54(1):1–14

Muller NZ, Mendelsohn RO (2009) Efficient pollution regulation: getting the prices right. Am Econ Rev 99(5):1714–1739

Muller NZ, Mendelsohn RO, Nordhaus WD (2011) Environmental accounting for pollution in the United States economy. Am Econ Rev (Forthcoming)

Nam KM, Waugh CJ, Paltsev S, et al (2013) Carbon co-benefits of tighter SO2 and NOx regulations in China. Global Environ Chang 23(6): 1648-1661

Nemet GF, Holloway T, Meier P (2010) Implications of incorporating air-quality co-benefits into climate change policymaking. Environ Res Lett 5

New Energy Externalities Development for Sustainability (NEEDS). Available at: <u>http://www.needs-project.org/</u>.

Oak Ridge National Laboratory (ORNL), Resources for the Future (RFF) (1995) External Costs and Benefits of Fuel Cycles (Reports 2–8). McGraw-Hill Utility Data Institute, Washington, 1994–1998

OECD (2012a) The Jobs Potential of a Shift Towards a Low-Carbon Economy. Organisation for Economic Co-operation and Development, Paris

OECD (2012b) Mortality Risk Valuation in Environment, Health and Transport Policies. Organisation for Economic Co-operation and Development, Paris

OECD (2014) The Cost of Air Pollution: Health Impacts of Road Transport. Organisation for Economic Co-operation and Development, Paris

Ottinger RL, Wooley D, Robinson N, et al (1990) Environmental costs of electricity. Oceana Publications, New York

Paltsev S, Reilly JM, Jacoby HD, et al (2005) The MIT Emissions Prediction and Policy Analysis (EPPA) Model: Version 4. MIT Joint Program on the Science and Policy of Global Change

Parry I (2014) Carbon Pricing: Good for You, Good for the Planet. iMFdirect – The International Monetary Fund's global economy forum

Parry I, Veung C, Heine D (2014) How Much Carbon Pricing is in Countries' Own Interests? The Critical Role of Co-Benefits. Social Science Research Network, Rochester, NY

Pearce D (1991) The role of carbon taxes in adjusting to global warming. Econ J 938–948

Pearce D, Bann C, Georgiou S (1992) The Social Cost of Fuel Cycles. HMSO, London.

Pittel K, Rübbelke DTG (2008) Climate policy and ancillary benefits: A survey and integration into the modelling of international negotiations on climate change. Ecol Econ 68:210–220. doi: 10.1016/j.ecolecon.2008.02.020

Preiss P, Klotz VE (2008) V1. 3: User's Manual & "Description of Updated and Extended Draft Tools for the Detailed Site-dependent Assessment of External Costs." Stuttgart: Institute of Energy Economics and the Rational Use of Energy, University of Stuttgart 63

Rabl A, Spadaro JV, Holland M (2014) How Much is Clean Air Worth? Calculating the Benefits of Pollution Control. Cambridge University Press

Rečka L, Ščasný M (2013) Environmental Regulation Impacts on the Czech Power System by the Dynamic Linear Optimisation Model Message. Politická ekonomie 2013:248–273.

Riekkola AK, Ahlgren EO, Söderholm P (2011) Ancillary benefits of climate policy in a small open economy: The case of Sweden. Energ Policy 39:4985–4998. doi: 10.1016/j.enpol.2011.06.015

Ščasný M, Rečka L (forthcoming) Ancillary benefits of Climate Change Mitigating Policies: Lessons from Slovak Energy Industry. Charles University Working Paper (mimeo), Prague

Scheraga JD, Leary NA (1993) Costs and side benefits of using energy taxes to mitigate global climate change. In *Proceedings of the Annual Conference on Taxation Held under the Auspices of the National Tax Association-Tax Institute of America* (pp. 133-138). National Tax Association-Tax Institute of America.

Somanathan E, Sterner T, Sugiyama T, et al (2014) National and Sub-national Policies and Institutions. Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change

Spadaro JV, Rabl A (2007) Global Health Impacts and Costs Due to Mercury Emissions. Risk Anal 28:603–613. doi: 10.1111/j.1539-6924.2008.01041.x

Stocker TF, Qin D, Plattner GK, et al (2013) IPCC, 2013: Summary for Policymakers. Climate Change 2013: The Physical Science Basis, Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change

Syri S, Amann M, Capros P, et al (2001) Low-CO2 energy pathways and regional air pollution in Europe. Energ Policy 29:871–884. doi: 10.1016/S0301-4215(01)00022-2

Van Harmelen T, Bakker J, de Vries B, et al (2002) Long-term reductions in costs of controlling regional air pollution in Europe due to climate policy. Environ Sci Policy 5:349–365. doi: 10.1016/S1462-9011(02)00043-6

Van Vuuren DP, Edmonds J, Kainuma M, et al (2011) The representative concentration pathways: an overview. Climatic Change 109:5–31

Van Vuuren DP, Kriegler E, O'Neill BC, et al (2014) A new scenario framework for Climate Change Research: scenario matrix architecture. Climatic Change 122:373–386

Viscusi WK (2010) The heterogeneity of the value of statistical life: Introduction and overview. J Risk Uncertainty 40:1–13.

Weinzettel J, Havránek M, Ščasný M (2012) A consumption-based indicator of the external costs of electricity. Ecol Indic 17:68–76.

West JJ, Smith SJ, Silva RA, et al (2013) Co-Benefits of Mitigating Global Greenhouse gas Emissions for Future Air Quality and Human Health. Nat Clim Change 3: 885–889

WHO Regional Office for Europe, OECD (2015) Health Economics and Air Pollution. Copenhagen: WHO Regional Office for Europe.

Appendix

Dellutent	11	Hard Coal			Ga	Gaseous fuels			avy Fuel	Oil	Biomass		
Pollulani	Unit	Value	Lower	Upper	Value	Lower	Upper	Value	Lower	Upper	Value	Lower	Upper
NMVOC	g/GJ	1	0.6	2.4	2.6	0.65	10.4	2.3	1.4	3.2	7.31	2.44	21.9
NOx	g/GJ	209	200	350	89	15	185	142	70	300	81	40	160
TSP	g/GJ	11.4	3	300	0.89	0.445	1.34	35.4	2	200	172	86	344
PM10	g/GJ	7.7	2	20	0.89	0.445	1.34	25.2	1.5	150	155	77	310
PM2.5	g/GJ	3.4	0.9	90	0.89	0.445	1.34	19.3	0.9	90	133	66	266
SOx	g/GJ	820	330	5000	0.281	0.169	0.393	495	146	1700	10.8	6.45	15.1
Cd	mg/GJ	0.9	0.627	1.46	0.25*	0.08*	0.75*	1.2	0.6	2.4	1.76	1.06	2.47
As	mg/GJ	7.1	5.04	11.8	120*	40*	360*	3.98	1.99	7.97	9.46	5.68	13.2
Ni	mg/GJ	4.9	3.44	8.03	0.51*	0.17*	1.53*	255	127	510	14.2	8.51	19.9
Pb	mg/GJ	7.3	5.16	12	1.5*	0.5*	4.5*	4.56	2.28	9.11	20.6	12.4	28.9
Hg	mg/GJ	1.4	1.02	2.38	100*	10*	1000*	0.341	0.17	0.682	1.51	0.903	2.11
Cr	mg/GJ	4.5	3.2	7.46	0.76*	0.25*	2.28*	2.55	1.27	5.1	9.03	5.42	12.6

Note: Lower and Upper values indicate the range of 95% confidence interval for given emission factor. Term * denotes factors expressed in g/GJ. Source: EMEP/EEA (2013).

Table A - 1. Emission factors for source category 1.A.1 - Public electricity and heat production.

Authors	Country	Scenarios	AB in 2005€/tCO₂	Pollutants	Impacts covered
Abt, 1999	USA	tax \$30, \$67 /tC*	2.3, 20	Criteria pollutants	Health – mortality and illness; Visibility and household soiling (materials damage)
Aunan, Aaheim, Seip, 2000	Hungary	Energy Conservation Program	148	TSP, SO ₂ , NOx, CO, VOC, CO ₂ , CH ₄ , N ₂ O, VOC	Health effects; damage on materials and vegetation. Annual health benefits \$648m, with a range of \$370 to \$1168m (Aaheim et al. 1997).
Barker and Rosendahl, 2000	Western Europe (19 regions)	Tax \$161/tC*	45	SO ₂ , NOx, PM10	Human and animal health and welfare, materials, buildings and other physical capital, vegetation
Boyd, Krutilla, Viscusi, 1995	USA	\$9/tC*	12	Pb, PM, SOx, SO ₄ , O ₃	Health, visibility
Brendemoen and Vennemo, 1994	Norway	Tax \$840/tC*	72	SO ₂ , NOx, CO, VOC,	Indirect: Health costs; lost recreational value from lakes and
				CO ₂ , CH ₄ , N ₂ O,	forests; corrosion
				Particulates	Direct: Traffic noise, road maintenance, congestion, accidents
Burtraw et al., 1999	USA	tax \$10, \$25, \$50 per ton C*	0.6 to 0.9	SO ₂ , NOx	Health
Burtraw et al., 2003	USA	\$25 carbon tax in the energy sector	12.5 – 14.5	NOx, PM10, TSP, SO2, sulfates	Health effects
Burtraw, Linn, Palmer, Paul, 2014	USA	cap-and-trade & tradable performance standards that reach 17% goal for 2020	34 – 44	SO ₂	Health effects based on damage factors in EPA (2011)
Cifuentes et al. 2000	Santiago, Chile	Energy efficiency	18	SO2, NOx, CO, NMHC, PM10, dust	Health
Dessus and O'Connor, 1999	Chile	Tax \$67, \$157, \$284 (10%, 20%, 30% C reduction)*	73, 74, 78	7 air pollutants	Health (morbidity, mortality, IQ effect)
Garbaccio, Ho, Jorgenson, 2000	China	Tax \$1/tC, \$2/tC*	15	PM10, SO ₂	Health
Grossman, Muller, O'Neill-Toy, 2011	USA	Warner-Lieberman bill (S.2191) of 2007	1-63	PM2.5, VOC, NOx, SO ₂ ,NH3 and O ₃	Health effects using APEEP model
Holland et al., 2011	EU27	2°C stabilization scenario	24	PM2.5, PM _{coarse} , SO ₂ , NOx	Health effects, effects on crops, building materials and ecosystems (ExternE).
Kiuila, Markandya, Ščasný,	Czech Republic	Full internalization of local	32 – 72	PM2.5, PM _{coarse} , SO ₂ ,	Health effects, effects on crops, building materials and
Tsuchimoto, forthcoming		external costs		NOx	ecosystems (ExternE).
Nemet, Holloway, Meier, 2010	review	NA	38 (1.6 – 152) 33 (developed)	NA	Health effects, various impact categories
Parry, Veung, Heine, 2014	20 top world-wide emitters	Nationally efficient carbon prices	44 (from -17 to 220) 65 (for coal)	PM2.5, SO ₂ , NOx	Health (intake fractions extrapolated from the average plant in China)
Rečka and Ščasný, 2013	Czech Republic	EU ETS till 2030	15	PM2.5, PM _{coarse} , SO ₂ , NOx	Health effects, effects on crops, building materials and ecosystems (ExternE).
Scheraga and Leary, 1993	USA	\$144/tC*	12	TSP, PM10, SOx, NOx, CO, VOC, CO ₂ , Pb	Health – morbidity and mortality
Ščasný and Rečka, forthcoming	Slovakia	17€/tCO2, -20% & -25% CO2 target	11	PM2.5, PM _{coarse} , SO ₂ , NOx	Health effects, effects on crops, building materials and ecosystems (ExternE).
West et al. 2013	14 world regions	NA	43 - 326	-	Health impacts, air quality model used

Note: * This information is based on OECD (2002) and tax is expressed in 1996 US\$. The ancillary benefits are recalculated in 2005 Euro by CPI and purchasing power standard rate. Table A - 2. Review of ancillary benefits per ton of CO₂, in €2005.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659 http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2015

ERM	1.2015	Elena Verdolini, Laura Diaz Anadon, Jiaqi Lu and Gregory F. Nemet: <u>The Effects of Expert Selection,</u> Elicitation Design, and R&D Assumptions on Experts' Estimates of the Euture Costs of Photovoltaics
CCSD	2 2015	lames Lennov and Ramiro Parrado: Canital-embodied Technologies in CGE Models
CCSD	3.2015	Claire Gavard and Diamel Kirat: Flexibility in the Market for International Carbon Credits and Price
		Dynamics Difference with European Allowances
CCSD	4.2015	Claire Gavard: Carbon Price and Wind Power Support in Denmark
CCSD	5.2015	Gunnar Luderer, Christoph Bertram, Katherine Calvin, Enrica De Cian and Elmar Kriegler: Implications of
		Weak Near-term Climate Policies on Long-term Mitigation Pathways
CCSD	6.2015	Francisco J. André and Luis M. de Castro: Incentives for Price Manipulation in Emission Permit Markets with
		Stackelberg Competition
CCSD	7.2015	C. Dionisio Pérez Blanco and Thomas Thaler: <u>Water Flows in the Economy. An Input-output Framework to</u>
		Assess Water Productivity in the Castile and León Region (Spain)
CCSD	8.2015	Carlos M. Gómez and C. Dionisio Pérez-Blanco: Simple Myths and Basic Maths about Greening Irrigation
CCSD	9.2015	Elorri Igos, Benedetto Rugani, Sameer Rege, Enrico Benetto, Laurent Drouet, Dan Zachary and Tom Haas:
		Integrated Environmental Assessment of Future Energy Scenarios Based on Economic Equilibrium Models
ERM	10.2015	Beatriz Martínez and Hipòlit Torró: <u>European Natural Gas Seasonal Effects on Futures Hedging</u>
CCSD	11.2015	Inge van den Bijgaart: The Unilateral Implementation of a Sustainable Growth Path with Directed Technical
		Change
CCSD	12.2015	Emanuele Massetti, Robert Mendelsohn and Shun Chonabayashi: <u>Using Degree Days to Value Farmland</u>
CCSD	13.2015	Stergios Athanassoglou: Revisiting Worst-case DEA for Composite Indicators
CCSD	14.2015	Francesco Silvestri and Stefano Ghinoi : Municipal Waste Selection and Disposal: Evidences from Lombardy
CCSD	15.2015	Loïc Berger: The Impact of Ambiguity Prudence on Insurance and Prevention
CCSD	16.2015	Vladimir Otrachshenko and Francesco Bosello: Identifying the Link Between Coastal Tourism and Marine
		Ecosystems in the Baltic, North Sea, and Mediterranean Countries
ERM	17.2015	Charles F. Mason, Lucija A. Muehlenbachs and Sheila M. Olmstead: <u>The Economics of Shale Gas</u>
		<u>Development</u>
ERM	18.2015	Anna Alberini and Charles Towe: Information v. Energy Efficiency Incentives: Evidence from Residential
		Electricity Consumption in Maryland
CCSD	19.2015	ZhongXiang Zhang: <u>Crossing the River by Feeling the Stones: The Case of Carbon Trading in China</u>
CCSD	20.2015	Petterson Molina Vale: <u>The Conservation versus Production Trade-off: Does Livestock Intensification</u>
		Increase Deforestation? The Case of the Brazilian Amazon
CCSD	21.2015	Valentina Bosetti, Melanie Heugues and Alessandro Tavoni: <u>Luring Others into Climate Action: Coalition</u>
CCCD	22 2015	Formation Games with Threshold and Spillover Effects
CCSD	22.2015	Francesco Bosello, Elisa Delpiazzo, and Fabio Eboli: <u>Macro-economic Impact Assessment of Future Changes</u>
CCCD	22 2015	In European Marine Ecosystem Services
CCSD	25.2015	Maryse Labriet, Laurent Drouet, Marc vielle, Richard Louiou, Amit Kanudia and Alain Haune: Assessment of
CCSD	24 2015	the Effectiveness of Global Climate Policies Osing Coupled Bottom-up and Top-Dom Models
CCSD	24.2013	Tacha la citological processo
	25 2015	Technological Frogress
CCJD	23.2013	Political Parties' Environmental Concern: An Empirical Analysis Recenting Op 01 NOL: The Determinants
CCSD	26 2015	Daniel Bodansky, Seth Hoedl, Cilbert Metcalf and Robert Stavins; Facilitating Linkage of Heterogeneous
CCJD	20.2015	Basional National and Sub-National Climate Delicies Through a Future International Agreement
CCSD	27 2015	Giannia, Vartonia, and Sub-Vartonia Camare Fonders model a future international Agreement
CCSD	27.2015	Bachires varias and Anastasios Acpapadeas. The Searce Externatives and the Management of Renewable
CCSD	28 2015	Todd Co.
CCDD	20.2015	Energy-Efficiency Gan and Its Implications for Climate Change Policy
CCSD	29,2015	Cristina Cattaneo and Emanuele Massetti: Migration and Climate Change in Rural Africa
FRM	30.2015	Simone Tagliapietra: The Euture of Renewable Energy in the Mediterranean. Translating Potential into
	00.2010	Reality
CCSD	31.2015	Ian Siegmeier, Linus Mattauch, Max Franks, David Klenert, Anselm Schultes and Ottmar Edenhofer: A Public
		Finance Perspective on Climate Policy: Six Interactions That May Enhance Welfare
CCSD	32.2015	Rever Gerlagh, Inge van den Bijgaart, Hans Nijland and Thomas Michielsen: Fiscal Policy and CO2 Emissions
		of New Passenger Cars in the EU
CCSD	33.2015	Marie-Laure Nauleau, Louis-Gaëtan Giraudet and Philippe Quirion: Energy Efficiency Policy with Price-
		quality Discrimination

CCSD	34.2015	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Athanasios Yannacopoulos: <u>Regulating the</u>
		Environmental Consequences of Preferences for Social Status within an Evolutionary Framework
CCSD	35.2015	Todd D. Gerarden, Richard G. Newell and Robert N. Stavins: <u>Assessing the Energy-efficiency Gap</u>
CCSD	36.2015	Lorenza Campagnolo and Fabio Eboli: <u>Implications of the 2030 EU Resource Efficiency Target on</u>
		Sustainable Development
CCSD	37.2015	Max Franks. Ottmar Edenhofer and Kai Lessmann: Why Finance Ministers Favor Carbon Taxes. Even if They
		Do not Take Climate Change into Account
CCSD	38 2015	ZhongXiang Zhang: Carbon Emissions Trading in China: The Evolution from Pilots to a Nationwide Scheme
CCSD	30.2015	David Carsis Loon Weather and Income Lassons from the Main European Paging
CCSD	10 2015	David Garcia-Leon. <u>Weather and monie. Lessons from the Affrain European Regions</u>
CCSD	40.2015	Jaroslav Myslak and C. D. Perez-Bianco: Partnersnips for Antoroable and Equitable Disaster Insurance
CCSD	41.2015	S. Surminski, J.C.J.H. Aerts, W.J.W. Botzen, P. Hudson, J. Mysiak and C. D. Perez-Blanco: <u>Reflections on the</u>
		<u>Current Debate on How to Link Flood Insurance and Disaster Risk Reduction in the European Union</u>
CCSD	42.2015	Erin Baker, Olaitan Olaleye and Lara Aleluia Reis: <u>Decision Frameworks and the Investment in R&D</u>
CCSD	43.2015	C. D. Pérez-Blanco and C. M. Gómez: <u>Revealing the Willingness to Pay for Income Insurance in Agriculture</u>
CCSD	44.2015	Banchongsan Charoensook: On the Interaction between Player Heterogeneity and Partner Heterogeneity in
		Two-way Flow Strict Nash Networks
CCSD	45.2015	Frin Baker, Valentina Bosetti, Laura Diaz Anadon, Max Henrion and Lara Aleluia Reis: Euture Costs of Key
CCDD	10.2010	Low-Carbon Energy Technologies: Harmonization and Aggregation of Energy Technology Expert Elicitation
		Data
CCCD	46 2015	
CCSD	46.2015	Sushanta Kumar Mahapatra and Keshab Chandra Ratha: <u>Sovereign States and Surging Water: Brahmaputra</u>
		River between China and India
CCSD	47.2015	Thomas Longden: <u>CO2</u> Intensity and the Importance of Country Level Differences: An Analysis of the
		Relationship Between per Capita Emissions and Population Density
CCSD	48.2015	Jussi Lintunen and Olli-Pekka Kuusela: Optimal Management of Markets for Bankable Emission Permits
CCSD	49.2015	Johannes Emmerling: Uncertainty and Natural Resources - Prudence Facing Doomsday
FRM	50 2015	Manfred Hafner and Simone Taglianietra: Turkish Stream: What Strategy for Europe?
ERM	51 2015	Thomas Sattick land Videsboard and Daniel Scholten: Can Ell's Decarbonization Agenda Break the State-
	51.2015	Comments Sattich, inga Testsbolid and Daniel Schöten. Car LO's Decarbonisation Agenda Dreak the State-
5514	50 004 5	Company Axis in the Power Sector?
ERM	52.2015	Alessandro Cologni, Elisa Scarpa and Francesco Giuseppe Sitzia: <u>Big Fish: Oil Markets and Speculation</u>
CCSD	53.2015	Joosung Lee: <u>Multilateral Bargaining in Networks: On the Prevalence of Inefficiencies</u>
CCSD	54.2015	P. Jean-Jacques Herings: <u>Equilibrium and Matching under Price Controls</u>
CCSD	55.2015	Nicole Tabasso: Diffusion of Multiple Information: On Information Resilience and the Power of Segregation
CCSD	56.2015	Diego Cerdeiro, Marcin Dziubinski and Sanjeev Goyal: Contagion Risk and Network Design
CCSD	57.2015	Yann Rébillé and Lionel Richefort: Networks of Many Public Goods with Non-Linear Best Replies
CCSD	58,2015	Achim Hagen and Klaus Eisenack: International Environmental Agreements with Asymmetric Countries:
0000	00.20.0	Climate Clubs vs. Global Cooperation
CCSD	50 2015	<u>Cimitate Guids vs. Giobal and Vincent Vennetalbaseb. Constitutions and Sasial Networks</u>
CCSD	59.2015	And invalueon, Nils Roeni and Vincent Vanneteibosch: <u>Constitutions and social Networks</u>
CCSD	60.2015	Adam N. Walker, Hans-Peter Welkard and Andries Kichter: <u>The Rise and Fall of the Great Fish Pact under</u>
		Endogenous Risk of Stock Collapse
CCSD	61.2015	Fabio Grazi and Henri Waisman: <u>Agglomeration, Urban Growth and Infrastructure in Global Climate Policy:</u>
		<u>A Dynamic CGE Approach</u>
CCSD	62.2015	Elorri Igos, Benedetto Rugani, Sameer Rege, Enrico Benetto, Laurent Drouet and Dan Zachary: Combination
		of Equilibrium Models and Hybrid Life Cycle-Input-Output Analysis to Predict the Environmental Impacts of
		Energy Policy Scenarios
CCSD	63 2015	Delayane B. Diaz: Estimating Clobal Damages from Sea Level Rise with the Coastal Impact and Adaptation
CCSD	00.2010	Model (CIAM)
CCCD	64 2015	<u>Midder (Chawr)</u>
CCSD	64.2015	Delavane B. Diaz: Integrated Assessment of Climate Catastrophes with Endogenous Uncertainty: Does the
		Risk of Ice Sheet Collapse Justify Precautionary Mitigation?
CCSD	65.2015	Jan Witajewski-Baltvilks, Elena Verdolini and Massimo Tavoni: <u>Bending The Learning Curve</u>
CCSD	66.2015	W. A. Brock and A. Xepapadeas: Modeling Coupled Climate, Ecosystems, and Economic Systems
CCSD	67.2015	Ricardo Nieva: The Coalitional Nash Bargaining Solution with Simultaneous Payoff Demands
CCSD	68.2015	Olivier Durand-Lasserve, Lorenza Campagnolo, Jean Chateau and Rob Dellink: Modelling of Distributional
		Impacts of Energy Subsidy Reforms: an Illustration with Indonesia
CCSD	69 2015	Simon Levin and Anastasios Xenanadeas: Transhoundary Canital and Pollution Flows and the Emergence of
CCSD	09.2013	Decise al Inequalities
CCCD	70 2015	Regional interdualities
CCSD	70.2015	Jaroslav Mysiak, Swenja Surminski, Annegret Thieken, Reinhard Mechler and Jeroen Aerts: Sendai Framework
		for Disaster Risk Reduction – Success or Warning Sign for Paris?
CCSD	71.2015	Massimo Tavoni and Detlef van Vuuren: <u>Regional Carbon Budgets: Do They Matter for Climate Policy?</u>
CCSD	72.2015	Francesco Vona, Giovanni Marin, Davide Consoli and David Popp: <u>Green Skills</u>
CCSD	73.2015	Luca Lambertini, Joanna Poyago-Theotoky and Alessandro Tampieri: <u>Cournot Competition and "Green"</u>
		Innovation: An Inverted-U Relationship
FS	74.2015	Michele Raitano and Francesco Vona: From the Cradle to the Grave: the Effect of Family Background on the
-		Career Path of Italian Men
FS	75 2015	Davide Carbonai and Carlo Drago: Positive Freedom in Networked Capitalism: An Empirical Analysis
	76 2015	Wai lin and Zhang Viang Zhangi Lavalling the Dlaving Eigld. On the Missing Data of Maturals Estimation
CCSD	70.2015	vver jin and zhongstiang zhang. Levening the Playing Field. On the Wissing Kole of Network Externality in
		Designing Kenewable Energy Technology Deployment Policies
ERM	//.2015	Niaz Bashiri Behmiri and Matteo Manera: <u>The Role of Outliers and Oil Price Shocks on Volatility of Metal</u>
		Prices
CCSD	78.2015	Jan Witajewski-Baltvilks, Elena Verdolini and Massimo Tavoni: Directed Technological Change and Energy
		Efficiency Improvements

ES	79.2015	David Cuberes and Rafael González-Val: <u>The Effect of the Spanish Reconquest on Iberian Cities</u>
CCSD	80.2015	Isabella Alloisio, Alessandro Antimiani, Simone Borghesi, Enrica De Cian, Maria Gaeta, Chiara Martini,
		Ramiro Parrado, Maria Cristina Tommasino, Elena Verdolini and Maria Rosa Virdis: <u>Pathways to Deep</u>
		Carbonization in Italy
CCSD	81.2015	Yonky Indrajaya, Edwin van der Werf, Hans-Peter Weikard, Frits Mohren and Ekko C. van Ierland: <u>The</u>
		Potential of REDD+ for Carbon Sequestration in Tropical Forests: Supply Curves for carbon storage for
		Kalimantan, Indonesia
ES	82.2015	Carlo Drago, Roberto Ricciuti, Paolo Santella: <u>An Attempt to Disperse the Italian Interlocking Directorship</u>
		Network: Analyzing the Effects of the 2011 Reform
CCSD	83.2015	Joseph E. Aldy: Policy Surveillance in the G-20 Fossil Fuel Subsidies Agreement: Lessons for Climate Policy
CCSD	84.2015	Milan Ščasný, Emanuele Massetti, Jan Melichar and Samuel Carrara: Quantifying the Ancillary Benefits of the
		Representative Concentration Pathways on Air Quality in Europe